IceTop: The surface component of IceCube

The IceCube Collaboration

R. Abbasi^{ab}, Y. Abdou^w, M. Ackermann^{ap}, J. Adams^p, J. A. Aguilar^v, M. Ahlers^{ab}, D. Altmannⁱ, K. Andeen^{ab}, J. Auffenberg^{ab}, X. Bai^{af,1}, M. Baker^{ab}, S. W. Barwick^x, V. Baum^{ac}, R. Bay^g, K. Beattie^h, J. J. Beatty^{r,s}, S. Bechet^m, J. Becker Tjus^j, K.-H. Becker^{ao}, M. Bell^{am}, M. L. Benabderrahmane^{ap}, S. BenZvi^{ab}, J. Berdermann^{ap}, P. Berghaus^{ap}, D. Berley^q, E. Bernardini^{ap}, D. Bertrand^m, D. Z. Besson^z, D. Bindig^{ao}, M. Bissok^a, E. Blaufuss^q, J. Blumenthal^a, D. J. Boersma^a, C. Bohm^{ai}, D. Boseⁿ, S. Böser^k, O. Botner^{an}, L. Brayeurⁿ, A. M. Brown^p, R. Bruijn^y, J. Brunner^{ap}, S. Buitinkⁿ, K. S. Caballero-Mora^{am}, M. Carson^w, J. Casey^e, M. Casierⁿ, D. Chirkin^{ab}, B. Christy^q, F. Clevermann^t, S. Cohen^y, D. F. Cowen^{am,al}, A. H. Cruz Silva^{ap}, M. Danninger^{ai}, J. Daughhetee^e, J. C. Davis^r, C. De Clercqⁿ, F. Descamps^w, P. Desiati^{ab}, G. de Vries-Uiterweerd^w, T. De Young^{am}, J. C. Díaz-Vélez^{ab}, J. Dreyer^j, J. P. Dumm^{ab}, M. Dunkman^{am}, R. Eagan^{am}, J. Eisch^{ab}, C. Elliott^{af}, R. W. Ellsworth^q, O. Engdegård^{an}, S. Euler^a, P. A. Evenson^{af}, O. Fadiran^{ab}, A. R. Fazely^f, A. Fedynitch^j, J. Feintzeig^{ab}, T. Feusels^w, K. Filimonov^g, C. Finley^{ai}, T. Fischer-Wasels^{ao}, S. Flis^{ai}, A. Franckowiak^k, R. Franke^{ap}, K. Frantzen^t, T. Fuchs^t, T. K. Gaisser^{af}, J. Gallagher^{aa}, L. Gerhardt^{h,g}, L. Gladstone^{ab}, T. Glüsenkamp^{ap}, A. Goldschmidt^h, J. A. Goodman^q, D. Góra^{ap}, D. Grant^u, A. Groß^{ae}, S. Grullon^{ab}, M. Gurtner^{ao}, C. Ha^{h,g}, A. Haj Ismail^w, A. Hallgren^{an}, F. Halzen^{ab}, K. Hanson^m, D. Heereman^m, P. Heimann^a, D. Heinen^a, K. Helbing^{ao}, R. Hellauer^q, S. Hickford^p, G. C. Hill^b, K. D. Hoffman^q, R. Hoffmann^{ao}, A. Homeier^k, K. Hoshina^{ab}, W. Huelsnitz^{q,2}, P. O. Hulth^{ai}, K. Hultqvist^{ai}, S. Hussain^{af}, A. Ishihara^o, E. Jacobi^{ap}, J. Jacobsen^{ab}, G. S. Japaridze^d, O. Jlelati^w, H. Johansson^{ai}, A. Kappesⁱ, T. Karg^{ap}, A. Karle^{ab}, J. Kiryluk^{aj}, F. Kislat^{ap}, J. Kläs^{ao}, S. R. Klein^{h,g}, S. Klepser^{ap}, J.-H. Köhne^t, G. Kohnen^{ad}, H. Kolanoski^{i,*}, L. Köpke^{ac}, C. Kopper^{ab}, S. Kopper^{ao}, D. J. Koskinen^{am}, M. Kowalski^k, M. Krasberg^{ab}, G. Kroll^{ac}, J. Kunnenⁿ, N. Kurahashi^{ab}, T. Kuwabara^{af}, M. Labareⁿ, K. Laihem^a, H. Landsman^{ab}, M. J. Larson^{ak}, R. Lauer^{ap}, M. Lesiak-Bzdak^{aj}, J. Lünemann^{ac}, J. Madsen^{ah}, R. Maruyama^{ab}, K. Mase^o, H. S. Matis^h, A. McDermott^{af}, F. McNally^{ab}, K. Meagher^q, M. Merck^{ab}, P. Mészáros^{al,am}, T. Meures^m, S. Miarecki^{h,g}, E. Middell^{ap}, N. Milke^t, J. Millerⁿ, L. Mohrmann^{ap}, T. Montaruli^{v,3}, R. Morse^{ab}, S. M. Movit^{al}, R. Nahnhauer^{ap}, U. Naumann^{ao}, P. Nießen^{af}, S. C. Nowicki^u, D. R. Nygren^h, A. Obertacke^{ao}, S. Odrowski^{ae}, A. Olivas^q, M. Olivo^j, A. O'Murchadha^m, S. Panknin^k, L. Paul^a, J. A. Pepper^{ak}, C. Pérez de los Heros^{an}, D. Pieloth^t, N. Pirk^{ap}, J. Posselt^{ao}, P. B. Price^g, G. T. Przybylski^h, L. Rädel^a, K. Rawlins^c, P. Redl^q, E. Resconi^{ae}, W. Rhode^t, M. Ribordy^y, M. Richman^q, B. Riedel^{ab}, J. P. Rodrigues^{ab}, J. Roth^{af}, F. Rothmaier^{ac}, C. Rott^r, C. Roucelle^h, T. Ruhe^t, D. Rutledge^{am}, B. Ruzybayev^{af}, D. Ryckbosch^w, S. M. Saba^j, T. Salameh^{am}, H.-G. Sander^{ac}, M. Santander^{ab}, S. Sarkar^{ag}, K. Schatto^{ac}, M. Scheel^a, F. Scheriau^t, T. Schmidt^q, M. Schmitz^t, S. Schoenen^a, S. Schöneberg^j, L. Schönherr^a, A. Schönwald^{ap}, A. Schukraft^a, L. Schulte^k, O. Schulz^{ae}, D. Seckel^{af}, S. H. Seo^{ai}, Y. Sestayo^{ae}, S. Seunarine¹, L. Shulman^{af}, M. W. E. Smith^{am}, M. Soiron^a, D. Soldin^{ao}, G. M. Spiczak^{ah}, C. Spiering^{ap}, M. Stamatikos^{r,4}, T. Stanev^{af}, A. Stasik^k, T. Stezelberger^h, R. G. Stokstad^h, A. Stößl^{ap}, S. Stoyanov^{af}, E. A. Strahlerⁿ, R. Ström^{an}, K-H. Sulanke^{ap}, G. W. Sullivan^q, H. Taavola^{an}, I. Taboada^e, A. Tamburro^{af}, S. Ter-Antonyan^f, S. Tilav^{af}, P. A. Toale^{ak}, S. Toscano^{ab}, M. Usner^k, D. van der Drift^{h,g}, N. van Eijndhovenⁿ, A. Van Overloop^w, J. van Santen^{ab}, M. Vehring^a, M. Voge^k, C. Walck^{ai}, T. Waldenmaierⁱ, M. Wallraff^a, M. Walter^{ap}, R. Wasserman^{am}, Ch. Weaver^{ab}, C. Wendt^{ab}, S. Westerhoff^{ab}, N. Whitehorn^{ab}, K. Wiebe^{ac}, C. H. Wiebusch^a, D. R. Williams^{ak}, H. Wissing^q, M. Wolf^{ai}, T. R. Wood^u, K. Woschnagg^g, C. Xu^{af}, D. L. Xu^{ak}, X. W. Xu^f, J. P. Yanez^{ap}, G. Yodh^x, S. Yoshida^o, P. Zarzhitsky^{ak},

J. Ziemann^t, A. Zilles^a, M. Zoll^{ai}

^aIII. Physikalisches Institut, RWTH Aachen University, D-52056 Aachen, Germany

^bSchool of Chemistry & Physics, University of Adelaide, Adelaide SA, 5005 Australia

^cDept. of Physics and Astronomy, University of Alaska Anchorage, 3211 Providence Dr., Anchorage, AK 99508, USA

^dCTSPS, Clark-Atlanta University, Atlanta, GA 30314, USA

^eSchool of Physics and Center for Relativistic Astrophysics, Georgia Institute of Technology, Atlanta, GA 30332, USA

- ^fDept. of Physics, Southern University, Baton Rouge, LA 70813, USA
- ^gDept. of Physics, University of California, Berkeley, CA 94720, USA

^hLawrence Berkeley National Laboratory, Berkeley, CA 94720, USA

ⁱInstitut für Physik, Humboldt-Universität zu Berlin, D-12489 Berlin, Germany

^jFakultät für Physik & Astronomie, Ruhr-Universität Bochum, D-44780 Bochum, Germany

^kPhysikalisches Institut, Universität Bonn, Nussallee 12, D-53115 Bonn, Germany ^lDept. of Physics, University of the West Indies, Cave Hill Campus, Bridgetown BB11000, Barbados

^mUniversité Libre de Bruxelles, Science Faculty CP230, B-1050 Brussels, Belgium

ⁿVrije Universiteit Brussel, Dienst ELEM, B-1050 Brussels, Belgium

^oDept. of Physics, Chiba University, Chiba 263-8522, Japan

^pDept. of Physics and Astronomy, University of Canterbury, Private Bag 4800, Christchurch, New Zealand

^qDept. of Physics, University of Maryland, College Park, MD 20742, USA

^rDept. of Physics and Center for Cosmology and Astro-Particle Physics, Ohio State University, Columbus, OH 43210, USA ^sDept. of Astronomy, Ohio State University, Columbus, OH 43210, USA ^tDept. of Physics, TU Dortmund University, D-44221 Dortmund, Germany "Dept. of Physics, University of Alberta, Edmonton, Alberta, Canada T6G 2G7 ^vDépartement de physique nucléaire et corpusculaire, Université de Genève, CH-1211 Genève, Switzerland v Dept. of Physics and Astronomy, University of Gent, B-9000 Gent, Belgium ^xDept. of Physics and Astronomy, University of California, Irvine, CA 92697, USA ^yLaboratory for High Energy Physics, École Polytechnique Fédérale, CH-1015 Lausanne, Switzerland ^zDept. of Physics and Astronomy, University of Kansas, Lawrence, KS 66045, USA ^{aa}Dept. of Astronomy, University of Wisconsin, Madison, WI 53706, USA ^{ab}Dept. of Physics and Wisconsin IceCube Particle Astrophysics Center. University of Wisconsin, Madison, WI 53706. USA ac Institute of Physics, University of Mainz, Staudinger Weg 7, D-55099 Mainz, Germany ad Université de Mons, 7000 Mons, Belgium ae T.U. Munich, D-85748 Garching, Germany ^{af}Bartol Research Institute and Department of Physics and Astronomy, University of Delaware, Newark, DE 19716, USA ^{ag}Dept. of Physics, University of Oxford, 1 Keble Road, Oxford OX1 3NP, UK ah Dept. of Physics, University of Wisconsin, River Falls, WI 54022, USA ai Oskar Klein Centre and Dept. of Physics, Stockholm University, SE-10691 Stockholm, Sweden ^{aj}Department of Physics and Astronomy, Stony Brook University, Stony Brook, NY 11794-3800, USA ak Dept. of Physics and Astronomy, University of Alabama, Tuscaloosa, AL 35487, USA ^{al}Dept. of Astronomy and Astrophysics, Pennsylvania State University, University Park, PA 16802, USA ^mDept. of Physics, Pennsylvania State University, University Park, PA 16802, USA ^{an}Dept. of Physics and Astronomy, Uppsala University, Box 516, S-75120 Uppsala, Sweden ao Dept. of Physics, University of Wuppertal, D-42119 Wuppertal, Germany ap DESY, D-15735 Zeuthen, Germany

Abstract

IceTop, the surface component of the IceCube Neutrino Observatory at the South Pole, is an air shower array with an area of 1 km². The detector allows a detailed exploration of the mass composition of primary cosmic rays in the energy range from about 100 TeV to 1 EeV by exploiting the correlation between the shower energy measured in IceTop and the energy deposited by muons in the deep ice. In this paper we report on the technical design, construction and installation, the trigger and data acquisition systems as well as the software framework for calibration, reconstruction and simulation. Finally the first experience from commissioning and operating the detector and the performance as an air shower detector will be discussed.

Keywords: IceCube, IceTop, cosmic rays, air shower, detector *PACS:* 95.55.Vj

Contents

1 Introduction			
2	Tech	unical design and deployment	6
	2.1	Design considerations	6
	2.2	General detector layout	6
	2.3	Tank design	8
	2.4	Deployment and commissioning	9
	2.5	Water freezing	11
	2.6	Survey of the tanks	13

^{*}Corresponding author

Email address: Hermann.Kolanoski@desy.de (H. Kolanoski)

¹Physics Department, South Dakota School of Mines and Technology, Rapid City, SD 57701, USA

²Los Alamos National Laboratory, Los Alamos, NM 87545, USA

³also Sezione INFN, Dipartimento di Fisica, I-70126, Bari, Italy

⁴NASA Goddard Space Flight Center, Greenbelt, MD 20771, USA

3	Front-end electronics3.1Digital Optical Module (DOM)3.2Signal capture and digitization3.3Pulse shapes and discriminator thresholds3.4PMT gains and saturation3.5Droop	13 13 14 16 17 18					
4	IceTop triggers and data acquisition4.1General scheme4.2Decisions at detector level4.3Data acquisition and processing	19 19 20 21					
5	Calibration5.1Calibration of DOM electronics and PMT5.2Tank calibration using atmospheric muons5.3Comparison to runs with muon taggers5.4SLC charge calibration	 23 23 24 27 29 					
6	Environmental conditions at South Pole6.1Overview6.2Atmosphere6.3Snow on tanks	30 30 31 31					
7	Signal processing and data preparation	32					
8	Air Shower Reconstruction8.1Time and charge distribution of air shower signals8.2Shower fluctuations8.3Likelihood fit8.4Shower size spectra and energy determination	33 34 34 35 37					
9	Simulation of air showers and the IceTop detector9.1Air shower simulation9.2Detector simulation9.3Definition of VEM units in simulation9.4Simulation production	38 38 38 40 41					
10	Performance 10.1 Tank properties and detector stability 10.2 Air shower detection in IceTop 10.3 Cosmic ray anisotropy 10.4 Primary composition 10.5 IceTop activity as veto for IceCube events 10.6 Calibration of IceCube with IceTop 10.7 Heliospheric physics with IceTop	42 42 44 44 46 47 47					
11	Summary and Outlook	47					
Ac	Acknowledgments 48						
Re	eferences	48					



Figure 1: The IceCube detector with its components DeepCore and IceTop in the final configuration (December 2010).

1. Introduction

The IceCube Neutrino Observatory at the geographic South Pole consists of a cubic-kilometer detector situated in the ice at a depth between 1450 m and 2450 m and a square-kilometer detector array at the surface. IceCube is primarily designed to measure neutrinos from below, thus using the Earth as a filter to discriminate against background induced by cosmic rays [1, 2, 3]. The detector employs optical sensors to detect light of charged particles generated by neutrinos in the ice or the Earth's crust. In addition, IceCube also includes a more densely instrumented part called DeepCore and an extensive air shower array on the surface called IceTop, both fully integrated into the IceCube data acquisition system. The IceTop array extends IceCube's capabilities for cosmic ray physics allowing the use of the full IceCube Observatory (Fig. 1) as a 3-dimensional array for the study of high-energy cosmic rays. Construction of IceCube, including the IceTop component, was completed in December 2010.

IceTop consists of Cherenkov tanks filled with clear ice that operate on the same principle as the water tanks of the Haverah Park experiment [4] and the Pierre Auger Observatory [5]. The tanks are arranged in pairs on the same, approximately 125 m, triangular grid as the vertical cables that carry the deep sensors of IceCube. The two tanks at each surface station are separated from each other by 10 m. Each tank contains two standard IceCube digital optical modules (DOMs, see Section 3.1). Air showers initiated in the atmosphere by cosmic rays are typically spread over a number of stations. The light generated in the tanks by the shower particles (electrons, photons, muons and hadrons) is a measure of the energy deposit of these particles in the tanks. The information from multiple stations is fitted to a model of the overall shower shape and intensity, called 'shower size' S, and the direction described by the zenith and azimuth angles θ, ϕ .

With the IceTop detector configuration at the 2835 m altitude of the South Pole surface, the threshold for efficient reconstruction of air showers that trigger 3 or more stations is approximately 300 TeV in most of the detector and approximately 100 TeV in a denser in-fill region. The geometrical acceptance of the combined surface and in-ice detectors is approximately 0.3 km² sr. This aperture will provide a useful rate of showers with energies up to few EeV. Thus, the energy range of IceCube as a cosmic-ray detector fully covers the knee region of the spectrum and extends to the energies where a transition from galactic cosmic rays to a population of extra-galactic particles may occur. The key to identifying a transition from one population to another is to know the nuclear composition of the primary cosmic radiation. General reviews of the status of cosmic ray physics in the region covered by IceTop can be found, for example, in [6] or in the cosmic ray review in [7]. A recent summary of composition measurements is given in [8].

Table 1: Acronyms used in this article.

ADC	Analog-to-Digital Converter
ATWD	Analog Transient Waveform Digitizer (front-end chip for signal recording)
DOM	Digital Optical Module (light sensor and readout electronics)
DOMHub	Computer in the IceCube Lab which receives the signals from the DOMs
fADC	fast Analog-to-Digital Converter (used in the DOM to record waveforms at a sampling rate of 40 MHz)
HG	high gain ('HG DOM': one of the two DOMs in a tank)
HLC	'Hard Local Coincidence' (in combination 'HLC hit' or 'HLC mode', describes an operation mode with full waveform readout)
HV	high voltage
IC	IceCube (in-ice part)
IT	ІсеТор
LC	local coincidence (of the two high-gain DOMs of a station)
LG	low gain ('LG DOM': one of the two DOMs in a tank)
MPE	multiple photoelectrons (used in 'MPE discriminator', see also SPE)
PE	photoelectron (charge which a single photoelectron generates and is seen by the elec- tronics, including PMT gain, used as charge unit)
PMT	photomultiplier tube
rms	root-mean-square
SLC	'soft local coincidence' (in combination 'SLC hit' or 'SLC mode', analogous to 'HLC mode' to describe the operation mode; despite the name, 'SLC' is not a 'coincidence')
SPE	single photoelectrons (used in 'SPE discriminator', see also MPE)
UTC	Coordinated Universal Time (reference time for all IceCube signals)
VEM	vertical equivalent muon (charge which a vertical muon generates in the IceTop DOM electronics of a tank, used as charge unit for IceTop signals)

The primary mass determination from extended air showers (EAS) is notoriously difficult because the measurements are indirect and have to rely on models for the hadronization processes. Observables sensitive to the primary mass composition are mainly the height of the shower maximum (measured through fluorescence or Cherenkov emission) and the number of muons in a shower. The highest energy muons stemming from the first interactions in the higher atmosphere are most closely correlated to the mass of the primary nucleus. IceCube, in combination with IceTop, offers the unique possibility to observe these muons, typically with initial energies above 500 GeV, in the deep ice in coincidence with the mostly electromagnetically deposited shower energy measured at the surface. This provides a particularly powerful method for the determination of the mass composition.

To mitigate the dependence on hadronization models, several alternative methods for studying mass composition with IceTop are available. Other mass sensitive observables are for example: the shower absorption in the atmosphere at different zenith angles, the number of dominantly low-energy muons in the surface detector, and other shower properties such as shower age (state of shower development) and shower front curvature.

The IceTop array has additionally been used to study high- p_T muons, PeV gamma rays and transient events, such as the radiation effects of solar flares. It also serves as a partial veto for the detection of downward-going neutrinos with IceCube and for direction calibration.

This paper is organized as follows: First, we describe the technical design, deployment and commissioning of IceTop in Section 2, followed by the description of the front-end electronics in Section 3 and the trigger and data acquisition systems in Section 4. Next, the calibration procedures and the environmental conditions which influence the shower detection are discussed in Section 5 and 6, respectively. The next part deals with the data handling and analysis, beginning with the signal processing and data preparation in Section 7 and continuing with the description of the air shower reconstruction in Section 8 and the simulation of air shower events in Section 9. Finally the performance of the detector is demonstrated in Section 10 with examples from finished and ongoing analyses. The paper ends with a summary and outlook. In all descriptions we concentrate on the current status of the detector and the software (spring 2012) and refer to previous configurations only if there were significant changes which matter for analyses using those data. Acronyms used in the following text are explained in Table 1.

2. Technical design and deployment

2.1. Design considerations

The IceTop air shower array was proposed as an addition to the IceCube neutrino telescope to extend the telescope's capabilities for cosmic ray physics and to partially veto the background of down-going muons. In order to realize this addition in the most cost effective way, technical developments and infrastructure of IceCube were employed as much as possible. These considerations defined the following guidelines for the design of IceTop:

- IceTop uses the same cables as laid out in trenches for the IceCube holes. That constrains IceTop's array size to 1 km² and the grid spacing to 125 m. Both together define the energy range from some 100 TeV to about 1 EeV.
- Whenever possible IceTop uses the same detector hardware, electronic readout, triggering scheme and data acquisition as in the deep-ice part of IceCube (henceforth referred to as 'in-ice IceCube').

Using the same Digital Optical Modules (DOMs) as in the deep ice for detection of Cherenkov light with the same readout scheme leads to the choice of ice as detector medium. The ice is produced in tanks similar to the water-filled tanks used in the Haverah Park experiment [4] and the Pierre Auger Observatory [5]. Such ice tanks are well suited for the detection of shower particles because a particle produces enough light so that signal fluctuations are not dominated by photon statistics. For example, in the finally produced tanks a vertically through-going muon produces approximately 125 photoelectrons. The choice of ice Cherenkov detectors saved design and development costs and also simplified integrating the IceTop signals into the in-ice data acquisition system.

Given the constraints on size, spacing, detector technology and basic components the following further considerations aim at optimization for air shower physics:

- Air showers have a huge range of different energy depositions in the tanks. To increase the dynamic range offered by the DOM electronics (see Section 3.1) each tank has two light detectors running at different gains. Having two DOMs in each tank also allows for an experimental determination of internal signal fluctuations by setting both DOMs to the same gain during special calibration runs.
- Each detector station consists of two tanks for the following reasons:
 - By requiring events to trigger both tanks of a station ('local coincidence') signals from extensive air showers can be distinguished from the high (typically 2 kHz) individual event rate generated mostly by low energy showers.
 - Signals seen in both tanks of a station can be compared on an event-by-event basis to give a measure of the intrinsic physical fluctuations in the shower front, both in timing and in amplitude.
 - IceTop can be divided into two very similar sub-arrays, each comprising one of the two tanks of each station, to measure the fluctuations of reconstruction parameters, such as shower front curvature, lateral distribution, core location accuracy, angular resolution.
 - Single-station events that trigger both tanks of a single, interior station without triggering adjacent stations could be exploited to extend the IceTop reach to lower energies. Such events have primary energies around 30 TeV and have typically at most one muon with sufficient energy to reach the deep detector.
- The size of the tanks and their wall reflectivity were optimized for large light output and short signal pulses. Both, large tank size and high reflectivity, yield high signals but long pulse lengths (in the final tank design the decay constant is about 30 ns).

2.2. General detector layout

The IceTop air shower array is located above the in-ice IceCube detector at a height of 2835 m above sea level, corresponding to an atmospheric depth of about 680 g/cm². It consists of 162 ice Cherenkov tanks, placed at 81 stations and distributed over an area of 1 km² on a grid with mean spacing of 125 m. Figure 2 shows a plan view of the final IceTop array. In the center of the array, three stations (numbered 79, 80, 81) are installed at intermediate positions. Together with the neighboring stations (numbered 26, 27, 36, 37, 46) they form an in-fill array for denser shower sampling. Each station comprises two cylindrical tanks, A and B, 10 m apart from each other. The tanks are embedded into the snow so that their top surface is level with the surrounding snow to minimize temperature variations and accumulation of drifting snow.

IceTop, as well as in-ice IceCube, took data already during the construction phase; the periods during which certain IceCube-IceTop configurations had been commissioned are reported in Table 2.

Table 2: List of the years when a certain configuration of IceCube (IC), IceTop (IT) became operational together with the numbers of the stations which were added in that year (fourth column). The numbers for IC include also the DeepCore strings. We will use abbreviations like IC79/IT73 for the constellation in 2010, for example. The fourth and following columns list stations using the numbers given in Fig. 2. The fifth column lists the tanks which have Tyvek liners (the others are coated with zirconia liner). However only the tanks installed in 2005 have higher reflectivities than the other tanks. The sixth column reports the DOMs which have the 'old' transformers with 43 Ω and a short time constant, see Sections 3.2, 3.5 and Table 4. The last two digits in this column (62 or 64) are the DOM numbers in a station according to Fig. 6.

Year	IC	IT	new IT stations	Tyvek liner	old transformer
	strings	stations			(43 Ω)
2005	1	4	21 29 30 39	all	all
2006	9	16	38 40 47-50 57-59 66 67 74	-	all
2007	22	26	46 55 56 64 65 71-73 77 78	-	(77,46,56)-62
					(77,46,71)-64
2008	40	40	44 45 52-54 60-63 68-70 75 76	-	(53,55)-62
					(52,55,62,68-70)-64
2009	59	59	2-6 9-13 17-20 26-28 36 37	-	37-64
2010	79	73	8 15 16 23-25 32-35 41-43 51	-	23-62
2011	86	81	1 7 14 22 31 79-81	79A, 80A, 81A/B	-



Figure 2: Locations of IceCube string holes and IceTop tanks with the IceCube Lab (ICL) in the center in the final configuration after 2010. The holes 81 to 86, belonging to DeepCore and not related to IceTop tanks, are not shown. IceTop stations are located next to IceCube strings (except for the in-fill station 81) and consist of two tanks, A and B. The irregularity of the array arises because tank locations were constrained by surface cabling and IceCube drilling operations. A denser in-fill array is formed by the stations 26, 27, 36, 37, 46, 79, 80, 81.



Figure 3: Tank dimensions in a cross-sectional view. Wall thicknesses are not to scale (see Table 3).

Table 3: Dimensions of IceTop tanks and assembly. The numbers given in units of m have tolerances of the order of 1 cm.

component	quantity	value
polyethylene tank	height	1.10 m
	wooden extension	0.20 m
	inner diameter	1.82 m
	wall thickness	6 mm
	zirconia liner	4 mm
ice	height	0.90 m
DOMs	distance between centers	0.58 m
perlite	thickness	0.40 m
outside tank	polystyrene below tank	$(100 \pm 2) \text{mm}$
	polyurethane foam around tank	(50 - 100) mm

2.3. Tank design

A schematic cross section of the IceTop detectors is shown in Fig. 3 (dimensions are reported in Table 3). The tanks are made of black, cross-linked polyethylene, 6 mm thick, 1.1 m high, with a 1.82 m inner diameter and are filled with ice to a height of 0.90 m. Most of the tanks have an integral diffusely reflective white liner made by dispersing zirconium dioxide powder⁵, referred to as zirconia, into High Density Polyethylene (HDPE) by extrusion⁶. The polyethylene, containing 6% zirconia by volume, is milled to fine white powder⁷ with average particle size 45 μ m. The tanks are produced by a rotational molding technique⁸ starting with molding the 6 mm thick, black outside layer and then covering it using the white powder to form a 4 mm thick layer. A final curing process leads to cross-linked bonds in the HDPE polymer structure which strengthens the tanks and finally determines the diffusive reflectivity of the liner. Eight tanks commissioned in 2005 and four tanks deployed in 2011 have Tyvek⁹ linings in form of bags loosely covering the tank walls (Table 2).

Figure 4 shows measurements of the diffuse reflectivity of tank liners as a function of the wavelength. The measurements were done outside of the tanks and usually in dry condition. For Tyvek also a comparison of reflectivities of dry and wet material from the 2005 tanks is shown in the plot. The difference between dry and

⁵Supplier: Stanford Materials, Irvine, CA 92618 U.S.A.

⁶Manufacturer: PlastiScience, LLC, Smyrna, DE 19977

⁷Manufacturer: Power King, Texas

⁸Manufacturer: PolyProcessing, Winchester VA

⁹Dupont brand of a synthetic textile made of high-density polyethylene fibers



Figure 4: Measurements of the wavelength dependence of diffuse reflectivity on samples from Tyvek bags ('uncoated' refers to the 2005 tanks) from tanks with zirconia liners. For more details see text.

wet Tyvek is relatively large because the water fills the Tyvek matrix which is for dry material filled by air. The reflectivity depends on the ratio of the refractive indices of filler and matrix which is difficult to predict for Tyvek under real conditions in ice-filled tanks. The Tyvek tanks commissioned in 2011 (Table 2) exhibit in ice optical properties which are resembling much more the properties of zirconia tanks according to pulse decay time and signal size (see Table 6). The reason for this unexpected behavior could not be fully traced back because several other conditions were different for the first tanks installed in 2005. For example, in 2005 the water came from the station, while later the water was prepared in the drill camp and additionally filtered. For zirconia tanks all reflectivity curves are quite similar as demonstrated by the two examples in Fig. 4. In contrast to the Tyvek liner the zirconia liner does not absorb water. However, the real conditions for the liner-ice boundary are not known in both cases. In the simulation (described in Section 9.2.1) the measured reflectivity curves are scaled to match the observed optical properties of the tanks, which are the pulse height and decay time (for typical single-particle pulses as measured with muons, see Section 3.3).

Figure 5 shows the tank assembly. Attached to the top of the tank is a wooden support structure (shown in brown in Fig. 5) that consists of $2'' \times 8''$ joists covered by hinged, plywood lids. The joists support the DOMs before the water freezes and the whole top serves as a structural platform to protect the detector after deployment. The hinged panels allow access for installing DOMs and let heat escape during the freezing process. The space between the ice surface and the lid is filled with expanded perlite (amorphous volcanic glass, expanded to low density with grain sizes of the order of 1 mm [9]) for thermal insulation and light protection (see Table 9 for composition).

The tank rests on a base of 10 cm thick rigid polystyrene insulation placed on top of a wooden pallet. The tank is mechanically strapped to the pallets with ropes. An enclosure for the Freeze Control Unit, described below, is mounted on the tank. On the right side of the Freeze Control Unit the drawing shows a tube into which poles carrying survey markers are inserted. Two vertical lines of temperature sensors are installed on the outer wall of the tank (not shown in Fig. 5) to monitor the freezing progress, one next to the Freeze Control Unit and one on the opposite side. There are 8 sensors per line in distances of about 15 cm, starting very near the tank bottom and ending about 15 cm above the water/ice level (measurements with these sensors are displayed in Fig. 9). The whole tank structure is integrated by applying expanding polyurethane foam around the tank from the pallet to the lid. The frame structure above the tank is only attached during the freezing process and carries sunshades to keep direct sunshine off the ice surface.

The assembly took place in the US and the completed units were transported to the South Pole. The sunshade frames were installed before deployment and removed after freezing was complete.

2.4. Deployment and commissioning

IceTop deployment was integrated with the in-ice drilling and deployment. Tank locations were laid out during the survey for the surface cables of IceCube. Stations consisting of two tanks, 10 m apart, were placed nominally 25 m from the top of the corresponding in-ice hole. Looking from the hole towards the tanks the left tank was labeled 'A' and the right one 'B' (Fig. 6), with the exception of station 48 where it was accidentally reversed. Tanks were set into trenches so the tops of the tanks were even with the snow surface to minimize drifting. After



Figure 5: Design rendering of an IceTop tank assembly before the outer foam insulation is applied.



Figure 6: Sketch of the orientation of tanks and DOMs. Indicated is the position of the tanks relative to the nearest IceCube hole. Between both tanks is the 'surface junction box' which connects the 64 DOMs of the IceCube string and the IceTop tanks to the cable running to the IceCube Lab. The string DOMs have the numbers 1 to 60, the IceTop DMs from 61 to 64. The IceTop DOM numbering scheme is shown in the sketch. The DOMs 61 and 63 have high gains and 62 and 64 low gains (see also Fig. 15).

the tanks were in position, the DOMs were mounted on hangers attached to the structural joists of the tanks (Fig. 5) with the PMT half facing down. The equator of the DOM was set at 90 cm above the bottom of the tank so that the photocathode portion of the DOM was submerged. The two DOMs in each tank were placed along a diameter of the tank symmetrically about its center with a separation of 58 cm between the centers of the two DOMs. Figure 7 shows a tank with its two DOMs installed in the ice just before the tank was filled with perlite insulation and closed.

A 'surface junction box' was placed between the two tanks of each station (Fig. 6) and the four IceTop DOMs as well as the corresponding in-ice cable with wires for its 60 in-ice DOMs were connected to it. The connection to the IceCube Lab (ICL) at the center of the array was made by a surface cable carrying the wires for the 64 DOMs at each hole/station. The order of deployment of the IceTop stations followed the trenching and laying of surface cables. The surface cable also carries wires to provide power and housekeeping signals for the Freeze Control



Figure 7: Photograph of a frozen tank ready to be closed. The degasser unit, which remains frozen at the bottom, is visible under 90 cm of ice in the lower left. The fuzziness in the outline of the degasser is mainly the result of imperfections in the surface of the ice.

Unit. At the end of the season, these wires were secured to the survey marker of each tank and are available for possible future use.

Since the Freeze Control Units had to be kept warm, the tanks could not be filled until the surface cables were connected in the IceCube Lab. When the surface cables were laid and connected to power, the Freeze Control Units were installed and immediately powered up. Soon after, the tanks were filled with water from the in-ice hot water drilling system (except for the tanks commissioned in 2005, see remark in Section 2.3). At the beginning of the drilling season the water contained as much as 4000 ppm of propylene glycol, which remained from the system protection for the winter. A filter assembly reduced the glycol content to a level of less than 10 ppm in the IceTop tanks. The filtered water was transported to each station in an insulated water reservoir mounted on a dedicated sled. The water was filled to a depth of 90 cm and the surface was maintained at that level until the freezing was complete.

Immediately after the tanks were filled, the Freeze Control Units were turned on with the tank lids closed for an initial two to three day period of cooling and degassing. During this period, sunshades were mounted on pre-installed frames that extended above the tanks (Fig. 5). The tanks were then opened and the freeze-in process lasting about 50 days began. The surface of the ice in the tanks had to be kept clear of snow to prevent an accumulation that would have insulated the tank and slowed or stopped the freeze process. After an initial period of operating the degasser nearly continuously, the duty cycle was decreased in steps to the minimal level needed to insure clear ice. After the freeze was complete the top of each tank (40 cm between the ice surface and the lid) was filled with expanded perlite insulation and the top closed. The sunshades were removed, and the DOMs could be turned on and commissioned. The snow was backfilled and groomed around the tanks. The Freeze Control Units were retrieved from the tanks and stored for future use.

2.5. Water freezing

A natural process that allows the water to freeze from the top to the bottom was adapted to make clear, crackfree ice. When ice freezes in a confined volume from the top down, two problems must be managed. As the ice front advances, air is trapped in the remaining water and would soon rise to saturation level and make bubbles if it were not removed. The expansion at the phase transition from water to ice also has to be dealt with.

The principle of the freezing operation is sketched in Fig. 8. The air is removed using a commercially available *contactor* (A in Fig. 8) consisting of a cluster of tiny tubes made out of a semi-permeable membrane passing through a plastic cylinder. A submersible pond pump (B in Fig. 8) with a five-micron filter to keep any sediment from fouling, forces water through these tubes. The contactor is kept at a modest vacuum (130 hPa). About half the dissolved air is removed in one pass with this setup. Except for the vacuum line and the power supplying the pump, the unit is self-contained. A negative pressure ensures any leaks are inward, and the resulting water is removed by a water tolerant vacuum pump. Tests showed that it is sufficient to run this assembly at the bottom of the tank. If the tank were perfectly insulated, the densest water (4 °C) would concentrate at the bottom and only



Figure 8: Schematic of the operation principle of the Freeze Control Unit: Initially the tank is filled until the water level spills over to the second container with the sump pump which pumps overflow water to the outside (third pipe). At the bottom of the tank, a pump pushes water through the degasser unit, keeping the water circulating until all the water is frozen.

the lower portion of the tank would be degassed. Even though the sides of the tank are insulated, there is enough thermal transfer so that a stable temperature gradient from top to bottom was never set up. The small amount of heat loss through the sides is sufficient to keep the water in the tank well mixed during the freezing process. The temperature of the entire volume, which remains near 0 $^{\circ}$ C, and the dissolved gas content are nearly independent of depth. With the pump operating at full capacity of 2 liter per minute, the entire 2500 liter volume of water in the tank can be processed about once per day.

Removal of the water as the ice front advances is also straightforward. The ice rapidly bonds to the sides of the tank with sufficient strength to form a plug. The pressure builds only high enough to force the water out through a pipe (Fig. 8) open at the bottom of the tank that extends into the insulated and heated Freeze Control Unit enclosure. The short horizontal section sets the initial water level in the tank. Inside the Freeze Control Unit, the water collects in a sump which is pumped out when full. Pumping water out rapidly when needed prevents the water from freezing and plugging that discharge hose which is exposed to about -30 °C temperatures. The Freeze Control Unit controls the circulation and vacuum pumps for the degasser, and the sump pump. Resistive heaters are used to keep the temperature inside the Freeze Control Unit at 30 °C.

Freeze history. The history of a freezing tank is illustrated in Fig. 9 depicting the depth profile of water/ice temperatures in the tank during the freezing process. The temperatures are measured through the line of temperature sensors mounted vertically on the outside. Initially the tank is empty so all of the sensors measure the ambient air temperature (here around -20 °C). Water is then added to the tank and all but one of the sensors now follow the water temperature. The exception is one sensor that is located above the surface level of the water. At this stage the tank lid is kept closed so the water can cool uniformly and the degassing unit can reduce the concentration of air in the water (see description above). When the water has reached equilibrium and is near the freezing point the lid is opened, causing an immediate drop in the 'free air' sensor (the highest in the sensor line) to near ambient level. The water begins to freeze from the top down. As the freezing front passes each sensor in turn the temperature drops below 0 °C. The thermal profile of the sensors near the surface follows the ambient temperature more closely than that of the deeper sensors as would be expected from the need for the heat to diffuse through the block of ice. Eventually all of the sensors fall below zero, but the thermal gradient remains high since there still is liquid water in the center of the tank even though the sides are completely frozen. At this stage water is also still being drained into the sump, further indication that the freeze is not complete. Finally, when the last water freezes the temperature gradient collapses and water evolution ceases. The freeze control unit is now disconnected, the perlite insulation placed on top of the tank, and the lid screwed shut. At this point the 'free air' sensor now joins the others as it is buried in the insulation and thermally better connected to the ice than it is to the outside air.



Figure 9: History of the freezing process as recorded by the temperature sensors which are mounted to the wall of the tank at different heights (8 sensors starting from the tank bottom to 15 cm above the water/ice level, see Section 2.3). In the plot the lowest curve, staying at about -20 °C, is measured by the highest sensor (above the water), the next curves following upwards are measured by the sequentially lower sensors.

2.6. Survey of the tanks

Surveyors determined the location of the center of each IceTop tank and the location of the survey marker on the tank. From those two measurements and the known tank parameters, the location of the center of each IceTop DOM is determined to an accuracy of 5 cm. The tops of the deep holes of IceCube were determined to a similar accuracy in the same coordinate system. Initial location of the deep DOMs was determined from the cable payout records and pressure sensor data. In a second stage LED light pulses generated in the DOMs (flashers) were used to determine relative vertical offsets of strings. Reconstructed muon data and flasher data are used to monitor for shearing or other changes over time, which have so far not been observed. Locations of deep DOMs are known to an accuracy of better than 1 m.

The IceTop tanks lie roughly on a tilted plane with the top of the tanks at z = 1944.54 m for the lowest point (tank 18A in Fig. 2) and at z = 1950.08 m for the highest (tank 74A) where z = 0 is defined in the center of IceCube.

3. Front-end electronics

The pivotal components of the detector are the Digital Optical Modules (DOMs). They detect the Cherenkov light produced by the shower particles in the ice of the tanks and convert these signals into digital information, which is sent to the central counting house, the IceCube Lab.

IceTop DOMs have the same hardware as all other DOMs in IceCube, but some details of the data acquisition system are different due the different physics requirements and different environmental conditions. For example, the pulses generated by air showers deliver in general more charge and thus the pulse thresholds are higher; another difference is that the IceTop DOMs are subject to the seasonal temperature changes while the temperature in the deep ice is extremely stable.

3.1. Digital Optical Module (DOM)

Each DOM [10] consists of a 10" Hamamatsu photomultiplier tube (PMT) [11] and electronic circuitry, contained inside a 33 cm glass pressure sphere, as shown in Fig. 10 a. The PMT is shielded from Earth's magnetic field by a mu-metal grid. The two DOMs of a tank are operated with different PMT gains, high gain (HG) and low gain (LG), to adapt the dynamical range to the extremely different signals in air showers (Table 4).

Data taking, triggering, digitization and communication with the IceCube Lab are controlled by an FPGA (Field Programmable Gate Array) with an on-chip CPU on the DOM mainboard. The CPU permits remote DOM



Figure 10: The IceCube Digital Optical Module (DOM): a) Schematic view of a DOM. b) Analog signal part of the block diagram of the DOM mainboard.

firmware updates. Timing of the DOM is controlled by a free-running 20 MHz oscillator which is regularly calibrated with the master clock in the IceCube Lab. In addition, each DOM mainboard is equipped with an LED for calibration, see Section 5.1 (there are other LEDs in the DOM which are used to study the optical properties of the deep ice but are not used in IceTop).

A detailed description of the Digital Optical Modules can be found in [10]. Some characteristic parameters for the IceTop DOM operation are summarized in Table 4.

3.2. Signal capture and digitization

The PMT signals are captured by the DOM electronics when the peak pulse voltage surpasses a programmable discriminator threshold. The relation between the pulse voltage, which the discriminator triggers on, and the pulse charge, which is the calibration quantity depends on the pulse shape and thus on the front-end electronics. Because of the importance for IceTop signal and threshold calibration we briefly describe the two electronic paths for pulse discrimination and recording.

The PMT's cathode is kept grounded with the high voltage (HV) applied to the anode from where the signals are read out. The signals are decoupled from the HV by a wide-band inductive transformer (also referred to as 'toroid') which is expected to have a higher reliability as compared to the more common capacitive decoupling. The effect of the droop caused by a transformer and its correction will be discussed below (Section 3.5).

The signal paths described here can be followed in the schematic given in Fig. 10 b. From the PMT's anode the signal passes the transformer and is then split into a path to the discriminator trigger and two paths to the signal recording electronics. A 'Pulser', feeding into the same paths, is used in the calibration procedure (Section 5.1).

If the signal passes the discriminator threshold, the PMT output is sampled by a custom-made integrated circuit called 'Analog Transient Waveform Digitizer' (ATWD) [13] in 3.33 ns wide bins for a total of 128 bins, corresponding to a total sampling time of about 427 ns. The signals on the ATWD path are delayed by 75 ns to allow for the trigger decision. Each ATWD has four channels, labeled ATWD0 to ATWD3, three of which are used for waveform recording. The fourth channel is not used for regular data taking but, depending on the running mode, other information can be recorded using a multiplexer (MUX in Fig. 10 b), see details in [10]. To minimize dead-time, each DOM is equipped with two ATWD chips which are used alternately.

The inputs to the channels 0, 1 and 2 are amplified by nominal gain factors 16, 2 and 0.25, respectively (Table 5). In the case of a discriminator trigger, the FPGA opens the input channels of one of the two ATWD chips and the three scaled signals are recorded synchronously. After the analog sampling, 128 Wilkinson ADCs digitize with 10 bit precision each of the 128 ATWD cells of a channel in parallel. The three channels are digitized sequentially, starting with the highest gain channel; channels with lower gain are only digitized if in the higher gain channels any bin is above 768 counts. This takes about 30 μ s per channel and up to 100 μ s for all three channels. The three different gain channels of the DOMs make up for an effective dynamic range of 16 bit. Measuring charges in units of PE, the charge yield of a single photoelectron, the effective dynamic range of the high-gain DOM in a 'standard' tank (Table 6) spans from about 0.15 PE to about 10 000 PE. The conversion from mV to PE is done by using the numbers in Table 6 (in this case for the 'standard' tank) which are specific to the multi-PE pulses seen in IceTop tanks and not to the pulse shape from single photoelectrons. The standard high-gain IceTop pulse

component	quantity	value
PMT	quantum efficiency at 390 nm (near maximum)	~ 25% [12]
	quantum efficiency, av. for 300 – 650 nm	~ 18% [11]
	cathode area	$\sim 550 {\rm cm}^2 [12]$
high voltage, gain		HG: 5×10^{6} at ~1250 V
		LG: 10^5 at ~750 V
	saturation current (50% from linearity)	HG: 140 mA
		LG: 50 mA
transformer	time constant (at -30 °C):	$1 - 2\mu$ s (old), 12μ s (new)
	front-end impedance	50Ω (old), 43Ω (new)

Table 5: Characteristics of the three ATWD readout channels (nominal or approximate values). The smallest digitization value of each channel is given in the third column as LSB = least significant bit.

channel	gain	LSB [mV]	saturation [mV]
0 (high gain)	16	0.125	~100
1 (medium gain)	2	1.0	~800
2 (low gain)	0.25	8.0	~7500

Table 6: IceTop pulse characteristics. The numbers are average or approximate values, actual values can deviate. The notations 'old' and 'standard' tanks refer to the tanks commissioned in 2005 and after 2005, respectively.

tank liner	'standard' tanks		'old' tanks	
PMT gain	HG	LG	HG	LG
rise time (10-90%)	10	10	10	10
decay time (observed) [ns]	31.0	31.0	43.5	43.5
decay time (input for simulation) [ns]	26.5	26.5	41.9	41.9
VEM charge [PE]	125	110	200	180
fraction of charge in peak bin	0.07	0.07	0.05	0.05
peak voltage of 1 VEM [mV]	110	2.2	100	2.0
peak voltage of 1 PE [mV]	0.88	0.02	0.5	0.01
total charge of 1 PE [mV] (sum of ATWD bins)	12.6	0.29	10.0	0.2
saturation in PE	8 000	125 000	14000	225000
saturation in VEM	64	1140	70	1250

has a peak voltage of 0.88 mV/PE corresponding to about 7% of the total charge in the peak bin (for the single photoelectron pulse shape these numbers are about 40% or 5 mV/PE in the peak bin).

In response to a trigger, the ATWDs are launched in discrete time steps of 25 ns, phase synchronized to the frequency-doubled 20 MHz DOM clock. The final time resolution is obtained from the position of the waveform in the ATWD with respect to the start time. The pulse time determination is described in Section 7.

In parallel, the pulses are digitized continuously by a commercial 10-bit fast ADC (fADC in Fig. 10b) at a sampling rate of 40 MHz. The fADC inputs are shaped with a time constant of 180 ns limiting the bandwidth to adjust to the lower sampling speed. In response to a trigger (Section 4) the downstream DOM electronics records fADC data for 6.4μ s. The fADC data have not yet been used in IceTop analyses but could be used for studies of signal tails extended in time.

If DOM local coincidence or certain other conditions as descibed in Section 4 are satisfied, the ATWD and fADC digitization streams are losslessly compressed and written to memory. If a discriminator has triggered and an ATWD is available, at least the sum of the ATWD bin contents, called chargestamp, and a timestamp is extracted and written to memory. The ATWD recording is not possible while a waveform acquisition is still in progress or



Figure 11: Typical IceTop waveforms. The most frequent shape has a steep rise, a single peak and an exponential tail. The waveform is here not droop corrected, but in panel a) the undershoot below the baseline after the pulse is very small, a feature of the 'new' transformers installed in DOM 09-61 (Table 3). An example for the 'old' transformer characteristics is shown in panel b) with a quite distinct undershoot. This DOM 21-63 belongs to the tanks installed in 2005 (Table 3) which exhibit longer decay times of the waveforms (Table 6). In the left plot the blue line depicts the slope of the leading edge with the extrapolation to the baseline defining the pulse time (red circle), see Section 7.

when both ATWDs are busy (see also the discussion on dead time in Section 4.2.4).

3.3. Pulse shapes and discriminator thresholds

As described above, the readout of signals is initiated if a pulse crosses a voltage threshold set by the DOM discriminator (upper path in the schematic in Fig. 10b). Any relation between voltages and charges depends on the pulse shape, which is determined by the tank properties and the electronics. Therefore we first discuss the typical shapes of tank pulses before turning to discriminator thresholds.

3.3.1. Pulse shapes and VEM units

Most of the pulses have a similar shape given mainly by the reflectivity of the tank walls because the spread of crossing times of the shower particles are short compared to the decay time due to reflections (Fig. 11). They can mostly be described by a superposition of standard pulses which is well represented by the signal of a single muon. The signal charge of a single muon is also used as a calibration standard for normalizing the signals of different tanks. The normalized charges are expressed in units of 'vertical equivalent muons' (VEM) determined by calibrating each DOM with muons (see Section 5).

The standard pulses are characterized by a fast rise of the leading edge, about 10 ns from 10% to 90%, and an exponential decay with a time constant of on average 31 ns for most of the tanks (Fig. 11 a) and 43.5 ns (Fig. 11 b) for the 8 tanks installed in 2005 (Section 2.3). If not otherwise mentioned we use as nominal pulse shapes those obtained for the tanks which were installed after 2005 (Table 6). In Table 6 two decay times are given: as observed and as put into the simulation. The difference is due to the electronic shaping which is applied also in the simulation as described in the next section.

3.3.2. Pulse shaping by the electronics

IceTop analyses are particularly dependent on a good simulation of pulse shapes to correctly reproduce the experimental thresholds: Unlike the case for in-ice DOMs where the thresholds are set on fractions of a PE, the multi-PE thresholds in IceTop are more affected by the actual pulse shape generated by many PE's.

For the simulation, pulse shaping is described by transfer functions for the different paths with which the PMT output has to be convoluted. The software simulating the DOMs [14] currently uses the following transfer function for the three ATWD channels and the discriminator path:

$$f(t) = \frac{1}{t\sqrt{\sigma\pi}} \exp\left(-\frac{(\ln(t) - \ln(t_m))^2}{2\sigma^2}\right)$$
(1)

with the parameters σ and t_m characterizing the general shape of the signal. The parameter σ depends on the transformer version (0.41 for the 'old', 0.35 for the 'new' transformers) and t_m is set for both to 13.53 ns. The function has a maximum at the time t_{max} which is related to the parameter t_m by:

$$t_{max} = t_m \, e^{-\sigma^2} \,. \tag{2}$$



Figure 12: Comparison of the transfer function currently used in the detector simulation, as described by the function (1) with the parameters given in the text, with the measured transfer functions of the ATWD and discriminator paths (for 'new' transformers).



Figure 13: Threshold monitoring plots for the high-gain DOM in tank 26A. a) The peak voltage of the first arriving pulse of the waveform shows a sharp rise at a threshold of 24 mV. b) The threshold in terms of charge is smeared out. The charge threshold can be defined by 50% point of the leading slope of the charge distribution as shown on the plot.

A detailed study [15] found that the ATWD and discriminator paths have similar, but not identical, transfer functions. Both can be described by (1), but the parameters σ and t_m are significantly different resulting in narrower pulses at the discriminator than at the ATWD. The shapes of the transfer functions with the measured parameters for both paths and those currently used in the simulation are displayed in Fig. 12. After convolution of the PMT pulses with the transfer functions, differences between both paths are reduced and a single set of simulation parameters gives a reasonable approximation.

3.3.3. Discriminator thresholds

Each DOM has two voltage sensitive discriminators, called SPE (single-PE) and MPE (multiple-PE), which are generally used for triggering waveform capture. Both discriminators feature programmable thresholds with ranges differing by a factor 10. In IceTop, both discriminators are used for multiple-PE thresholds. In high-gain DOMs, the MPE discriminators are used for triggering on air showers with thresholds of about 20 mV corresponding to a signal charge of about 23 PE for a standard pulse (see Fig. 13), while the low-gain DOMs use the SPE discriminators for the same purpose with thresholds of about 4 mV corresponding to 270 PE [16]. With these thresholds, the single DOM rates are about 1600 Hz.

The SPE discriminators of the high-gain DOMs are used to record 'scaler rates' at various thresholds (from about 0.5 PE to 30 PE) without triggering waveform capture. An example for the threshold dependence of rates is shown in Fig. 14. The scaler rates are monitored continuously for solar flares, gamma ray bursts and other transient events.

3.4. PMT gains and saturation

The two DOMs in each tank are operated at different PMT gains (Table 4): The high-gain DOMs are nominally set to a gain of 5×10^6 , half the standard gain for in-ice DOMs. At this gain the saturation current (deviation by



Figure 14: Count rate as a function of discriminator setting of different DOMs. Rates are averaged over the interval from 17 February 22:00 UT to 18 February 02:00, which defines the "base count rate" for a study of a Forbush decrease observed in February, 2011 [17]. The line shows the expected values based on a simulation [18].

50% from linearity) is about 140 mA corresponding to about 8 000 PE for the standard IceTop pulses (Table 6). The low-gain DOMs are set to a gain of 10^5 and have a saturation current of 50 mA which corresponds to about 125 000 PE for standard IceTop pulses. Although the gain of the low-gain DOMs is 50 times smaller than of the high-gain DOMs, the number of photoelectrons at saturation level increases only by a factor of 15. The available 16 bit dynamic range of the ATWDs is about 10 000 PE for high-gain DOMs and 450 000 PE for low-gain DOMs (standard pulses), thus covering the non-saturated PMT range. The change from high-gain DOM to low-gain DOM signals is fixed to a much lower PE value, corresponding to less deviation from linearity (Section 5.2.5). Since the pulse shape stays very similar in a wide range (except for very low pulses and beyond saturation) the saturation can also be expressed in terms of VEM (for the 'standard' pulse, see Table 6). The low-gain DOM saturation determines the IceTop tank saturation which is typically of the order of 1000 VEM.

The implementation of saturation effects in the simulation is described in Section 9.2.

3.5. Droop

The transformer described in Section 3.2 leads to a characteristic high-pass frequency response for all signals. While the time constant is large compared to IceTop pulse widths, the later parts of each waveform show a 'droop' in amplitude, followed by undershoot below the baseline after the pulse ends. The pedestals of the ATWD and fADC are set to about 10% of the maximum scale, to permit waveforms with below-baseline excursions (that effectively also reduces the available 10 bit range).

Choosing the time constants of the transformer requires a compromise between loss of precision in high-frequency signals, and large droop after large signal amplitudes. The first DOMs produced had transformers with short time constants of about $1 \mu s$ at $-30 \degree C$ leading to a strong droop effect (Fig. 11 b), whereas DOMs produced later (since 2005) have larger time constants of about $12 \mu s$ at $-30 \degree C$ with reduced droop (Fig. 11 a). In IceTop, 79 of the 324 DOMs or 27% of all DOMs have transformers with large droop, see Table 2.

The droop effect can be described empirically by an impulse response with two time constants $\tau_{1,2}$ [19],

$$\delta(t) \to \delta(t) - N \left((1 - f) e^{-t/\tau_1} + f e^{-t/\tau_2} \right), \tag{3}$$

where N is a normalization constant. The formula is used in a discretized form for simulation and corrections of measured pulses (see Section 7). The droop constants f, τ_1 , and τ_2 have been measured in the laboratory as a function of DOM temperature. The temperature dependence is (T in °C)

$$\tau_1(T) = A + \frac{B}{1 + e^{-T/C}}$$

$$\tau_2(T) = 0.75 \tau_1(T)$$
(4)

Using a single-tau approximation by setting f = 0 the parameters are on average:

$$A = 0.4\,\mu\text{s}, \quad B = 5.0\,\mu\text{s}, \quad C = 16\,^{\circ}\text{C} \quad \text{(old transformers)} \\ A = 1.0\,\mu\text{s}, \quad B = 50.0\,\mu\text{s}, \quad C = 25\,^{\circ}\text{C} \quad \text{(new transformers)}$$
(5)

In the IceTop tanks the DOM mainboard temperatures vary between about -20 °C and -40 °C throughout the year (in the deep ice the temperatures are stable), yielding for the new transformers τ values between about 9 and 17 μ s. This has a major impact on the signal extraction and requires a continuous monitoring of the DOM temperatures. The inversion of the formula (3) is used to remove the droop in the data, see Section 7.

4. IceTop triggers and data acquisition

4.1. General scheme

In normal operation, the DOMs act as autonomous data collectors, initiating the capture and digitization of waveforms in response to one of many triggering conditions (generated either internally to DOM or from within a full IceTop station). A DOM is capable of sending a variety of data records, depending on selected operation modes, to the IceCube Lab for consideration in detector-wide triggers and event construction or for calibration and monitoring purposes. IceTop data included in these triggered detector-wide readouts, known as an 'event', are available to a suite of filter algorithms, selecting the most interesting events for immediate transmission north via a dedicated satellite link.

Trigger scheme. To select physically interesting events and to save bandwidth for data transfer IceCube has a multi-level trigger system including both hardware and software decisions. In the following an overview of the different 'triggers' is given, which also serves to define the terminology:

- **Discriminator Trigger:** Normally the capture of the waveform by an ATWD is initiated by the FPGA, when a pulse passes a discriminator threshold; autonomous within a DOM; described in Section 3.3.
- **DOM Launch:** The initiation of the process of capture and digitization of waveforms and data preparation for transmission organized by the FPGA according to the operation mode. Normally, a DOM is launched by the discriminator trigger, but the trigger can also be generated internally by the FPGA, for example for a baseline determination (Section 4.2.3). To be launched the DOM has to be in the 'ready' state (see the discussion of dead times below).
- Hit: The basic information recorded upon a DOM launch is referred to as a 'hit'.
- **Local Coincidence (LC):** Each DOM in IceTop is connected by a network of cable connections to neighboring DOMs in the same station. The presence of a logical level set by the discriminator trigger in neighboring DOMs, the so-called Local Coincidence condition, can be used to set the content of hits sent to the IceCube Lab. Described in Section 4.2.1.
- **Software trigger:** A series of software algorithms formed in the IceCube Lab using DOM Launches satisfying the LC condition determine when data from the entire IceCube detector should be recorded into events.
- **Online filter:** A subset of events are selected for immediate transfer north by satellite link using a set of filtering algorithms that select the physically most interesting events for later detailed analysis. All events are archived on magnetic tape storage.

Operation modes. How much information a hit contains when being transferred to the IceCube Lab, is determined by the operation modes for the readout of DOMs. Two different modes were foreseen for IceCube [10]:

SLC mode: In the 'soft local coincidence' mode data from all launched DOMs are transferred to the IceCube Lab. The data format of a hit contains a header with coarse information about the hit and in addition the full waveform data for 'HLC hits', that are hits with a fulfilled LC condition ('LC tag'). For hits which have no LC tag ('SLC hits') only the header with the coarse information is transmitted. For IceTop the coarse information is a timestamp and the integrated ATWD charge. See Section 4.2.2.

HLC mode: In the 'hard local coincidence' mode only those DOMs transfer data which have an LC tag.

In both modes additional conditions can lead to full waveform transfer, such as in case of the calibration triggers described below in Section 4.2.3. The baseline-operation of IceCube is the SLC mode [10] which, however, was not applied before 2009. For IceTop the standard SLC mode has been modified so that the ATWD digitization is not aborted if there is no LC tag because the integrated ATWD charge is always transmitted in the SLC mode (for in-ice DOMs only fADC information is used for the SLC transmission so that the ATWD digitization can be aborted if there is no LC tag).



Figure 15: Schematic of the Local Coincidence configuration of high-gain (HG) and low-gain (LG) DOMs in an IceTop station. The coincidence requirements are described in the text. The DOM numbering is explained in Fig. 6.

4.2. Decisions at detector level

4.2.1. Local Coincidence

For standard data taking the full-waveform readout of a DOM is initiated if a local coincidence (LC) of a station is fulfilled. This results in a single station trigger rate of about 30 Hz compared to about 1600 Hz of a single high-gain DOM at a threshold of about 0.1 VEM. The single DOM rate is dominated by small showers with a good fraction caused by single muons passing a tank (see Section 5.2.2).

To establish a local coincidence the high-gain and low-gain DOMs of a station are interconnected via cables as shown in Fig. 15. The following simplified description of the local coincidence decision focuses on the IceTop applications (details can be found in [10]). If a high-gain DOM passes the discriminator threshold a logical signal is sent to both DOMs in the other tank, which is used to define a 1 μ s coincidence time window. In the other tank, a DOM of either gain is read out if it had a discriminator trigger within this time window. Thus, in the normal configuration, station data always consist of signals from both high-gain DOMs and optionally additional signals from low-gain DOMs. However, due to the fact that the LC condition is only based on the discriminator trigger, dead time (see below) can cause data from a high-gain DOM to be missing in the rare cases where both ATWDs are busy.

The connections between high-gain and low-gain DOMs and an additional, normally unused connection between the two low-gain DOMs (not shown in Fig. 15) add also redundancy for the case that one or both of the high-gain DOMs stops functioning. In such an event low-gain DOMs can be switched to high-gain and an LC connection can be reestablished.

A fulfilled LC condition will result in storing the digitized waveforms into the local DOM memory from where it will be transfered to the IceCube Lab as described in Section 4.3.1.

4.2.2. Soft Local Coincidence

As explained above, 'SLC mode' refers to an operation mode which reads out every DOM for which a signal passed the discriminator threshold, at minimum recording the sparse 'SLC hit' information of integrated charge and timestamp. Normally, only HLC hits include the full waveform information. Including SLC hits in the analysis is useful for detecting single muons in showers where the electromagnetic component has been absorbed (low energies, outer region of showers, inclined showers) as well as for vetoing air shower background for in-ice studies.

4.2.3. Calibration launches

Under certain conditions full waveform readout can be initiated without the LC condition being fulfilled. Currently there are two such special DOM launches, both serve calibration or monitoring purposes. VEMCal launch: This launch of single high-gain DOMs collects single muon hits for calibration of the VEM unit. It is described in Section 5.2.2. Beacon launch: In order to determine the ATWD and fADC baselines from 'empty events' all DOMs are launched at a rate of 1 Hz without requiring that a discriminator trigger has fired.

4.2.4. Dead time and unusual hit patterns

In order to achieve a DOM launch the DOM must be in the "ready" state. Several conditions may cause a DOM to be not ready. If the DOM got a valid LC the DOM is in a "busy" state for $6.4 \mu s$ (the time the fADC records the waveform). If the DOM did not get a valid LC a new DOM launch is disabled until the DOM knows it did not receive a valid LC. This time is currently set to $2.5 \mu s$ for IceTop. For a launch rate of 1600 Hz, this creates a dead time probability of 0.4%. After that time the DOM is reenabled if there is a free ATWD, but if both ATWDs are busy the DOM remains disabled until an ATWD becomes available. For IceTop the SLC mode requires always the ATWD digitization to be completed (see the definiton of the SLC mode above), which takes about $30 \mu s$ (Section 3.2). Assuming the 1600 Hz of launches is uncorrelated, there is about 5% probability of overlapping the



Figure 16: Schematic of the data flow and the software components of IceCube data acquisition system. The data from the DOMs are collected by the StringHub modules (residing in the DOMHub computers) which send data to the trigger modules. Upon a global trigger the Event Builder gathers the corresponding data of the trigger window and sends the data to the online processing and filtering system PnF.

processing from two DOM launches, and a dead time probability of about 2.5% that a discriminator crossing will not result in a DOM launch. When operating in HLC mode, if there is no LC condition, the ATWD launch can be aborted, in which case the ATWD in question can be re-enabled after about $6.5 \,\mu$ s (the LC decision time plus about $4 \,\mu$ s), and the LC-coincidence dead time is reduced to 0.5%. In HLC mode, other cases occur under unusual circumstances; for example, if the high-gain DOM in one tank (say, A) and the low-gain DOM in the other tank (B) are the only DOMs with a discriminator trigger, then a full report for the low-gain DOM will be generated, but no report will be made for the DOM launch in tank A. This might occur if a single low energy electron occurs in proximity to the low-gain DOM in tank B, so that the light signal in the high-gain DOM is below threshold.

4.3. Data acquisition and processing

An overview of the data acquisition system of all IceCube components is given in Fig. 16. In the following we explain the features relevant to IceTop operation.

4.3.1. Data transfer to the IceCube Lab

Each IceTop DOM is connected to a computer, called DOMHub (Fig. 16), by a single twisted pair of wires. This differs from the setup for in-ice DOMs, where the lower counting rate allows each twisted pair to carry the signals from two adjacent DOMs. The DOMHubs are standard personal computers equipped with PCI digital optical module readout (DOR) cards. Each DOMHub accommodates eight IceTop stations (32 DOMs).

A fulfilled DOM launch condition, usually the LC condition, will result in storing the digitized waveforms of the highest-gain ATWD channel which is not saturated (the lowest-gain channel if all are saturated) and of the fADC into the local DOM memory. In regular intervals the DOMHubs request these data to be transferred. In SLC mode, time and integrated charge information is always transmitted and for HLC hits (and some special DOM launches, see Section 4.2.3) in addition the full waveform is transmitted.

4.3.2. Data processing in the IceCube Lab

In the DOMHubs the data are prepared for further processing by the StringHub modules. First the timestamps of each hit, which are calibrated to a common time reference by the RAPcal system (Section 5.1), are converted to UTC time. The hits are then time sorted and grouped together within time windows of a few microseconds. Subsequently the HLC hits are sent to the trigger units of the in-ice detectors and of IceTop where trigger algorithms are applied (see below). The triggers define readout windows, which differ for the different sub-detectors and triggers. In IceTop the window for the normal air shower events is $\pm 10\,\mu$ s before the first and after the last hits which produced the trigger. The IceTop calibration trigger has a window of $\pm 1\,\mu$ s. The global trigger merges all overlapping individual trigger windows for all sub-detectors. Based on this global readout window, the event builder collects all launches from all sub-detectors that happen during the global trigger window.

In the common IceCube event builder the data selected by the global trigger are transformed into the DAQ data format, sent to the online processing and filtering (PnF) and are stored on tapes. The tapes can only be shipped to

Table 7: Summary of IceTop rates.

name	condition	rate [Hz]	
	hit rates		
DOM (single)	DOM launch	1600	
HLC (all DOMs)	LC fulfilled	3000	
	triggers		
IceTopSMT	\geq 6 HLC hits	30	
IceTopMinBias	HLC prescale 10 ⁴	0.3	
IceTopCalibration	VEMCal launch	30	

the north during the Antarctic summer. Therefore part of the data, as defined by the filters, are sent via satellite transmission. In addition, the data for monitoring transient events (Section 4.3.5) are also sent via satellite.

4.3.3. IceTop trigger

The current basic trigger for IceTop air shower physics is the IceTop Simple Multiplicity Trigger (IceTopSMT) which requires 6 HLC hits (usually from 3 stations) in IceTop DOMs within 6μ s:

IceTopSMT :
$$\geq 6$$
 HLC hits. (6)

The readout window starts $10\,\mu$ s before the trigger window and lasts until $10\,\mu$ s after the last of the 6 hits. Upon an IceTop trigger the whole detector is read out and similarly, when an in-ice HLC trigger is formed, the previous interval is recorded to capture any activity in IceTop.

In addition a 'minimum bias' trigger, IceTopMinBias, triggers on any HLC launch in IceTop with a 'prescale' factor of 10⁴, that means only every 10⁴th HLC launch triggers.

All IceTop VEMCal launches (Section 4.2.3) are selected by the trigger IceTopCalibration. The treatment of calibration triggers is described in Section 5.

A summary of various trigger rates is given in Table 7.

4.3.4. IceTop online filter

For satellite transmission the data have to be filtered for data reduction. The filters, summarized for IceTop in Table 8, select events according to various physics cases or for monitoring purposes. For some filter classes only a certain fraction of events is transmitted (prescaling).

The basic IceTop filter. The most important IceTop filter, called IceTopSTA3, is based on events triggered by the simple multiplicity trigger, IceTopSMT, and requires that at least 3 stations have an HLC hit. During the construction phase events with less than 8 stations were prescaled for satellite transmission. From 2012 on all events with 3 or more stations are transmitted without prescale, though with partly only condensed information employing the so-called SuperDST format (see below). The three-station condition has an energy threshold of about 300 TeV.

To allow adjusting to different transfer and/or analysis modes the events passing the IceTopSTA3 filter are further broken down by the IceTopSTA5 and IceTop_InFill_STA3 filters. The IceTopSTA5 filter requires at least 5 stations to be hit. The IceTop_InFill_STA3 filter, which aims to select events with energies as low as 100 TeV, requires at least 3 stations from the infill tanks (Section 2.2) to be hit.

There is an additional IceTop filter, called InIceSMT_IceTopCoin (8), which is independent of the IceTopSMT trigger. It requires the in-ice simple multiplicity trigger to be fulfilled and at least one HLC hit in IceTop. Events which pass the InIceSMT_IceTopCoin filter consist of coincident events triggering in-ice IceCube with mostly just one IceTop station. These events will be used for vetoing high energy inclined cosmic ray showers in addition to being used to test and calibrate the entire IceCube detector (see Section 10.6).

4.3.5. Data formats and satellite transfer

For all events passing the filter conditions IceTopSTA3 or InIceSMT_IceTopCoin the total charge and the leading edge time of all DOM launches is recorded in a condensed format called SuperDST and transmitted via satellite. Using a logarithmic representation the IceTop charges, which span 5 orders of magnitude, have to be encoded by 14 bit to ensure a precision of about 3×10^{-3} throughout the whole dynamic range. A total of 5 byte

Table 8: IceTop filters with their inclusive rates. The rates for full waveform transmission via satellite after associated prescaling (4th column) and the corresponding data loads are reported in the last 2 columns. All events including the prescaled ones are transmitted in the condensed SuperDST format as described in the text.

filter	condition	rate	full wavef	form subset	satellite data
		[Hz]	prescale	rate [Hz]	[GB/day]
IceTopSTA3	IceTopSMT & # Stations ≥ 3	22.3	10	2.2	0.9
IceTopSTA5	IceTopSMT & # Stations ≥ 5	6.1	1	6.1	3.1
IceTop_InFill_STA3	IceTopSMT & # In-fill Stations ≥ 3	3.7	1	3.7	1.3
InIceSMT_IceTopCoin	InIceSMT & # IceTop HLCs ≥ 1	100	100	1.0	0.5

per DOM is necessary for encoding charge, time, station number, DOM number and LC bit. For the observed mean number of 22 launches per event the average event size yields 110 bytes. The estimated IceTop contribution to the data rate for all events in SuperDST format is less than 500 MB per day.

In addition to the transmission in SuperDST format, the full waveform will be transmitted for selected or prescaled events according to the scheme summarized in Table 8. For the IceTopSTA5 and IceTop_InFill_STA3 filters all events will have the full waveform information, while for the others only some fraction of events will have full waveform information for control studies.

Transient rate monitoring. In 1 s intervals, the rates of the SPE and MPE discriminators (see Section 3.3.3) as well as of the Local Coincidences are written out for monitoring and detection of transient events. These so-called scaler rates are also transmitted for each run via satellite. There is a delay of typically a day until the data are accessible in the North.

5. Calibration

The tank signals are calibrated in two steps: First, the PMT and the DOM electronics are calibrated in order to obtain signal charges in units of photoelectrons and a time reference which is the same for all DOMs in all parts of IceCube. This part is common to all DOMs in IceCube. The second step is the calibration of the tank signal charges in units "vertical equivalent muon" (VEM) which is the signal a vertical energetic muon generates in a tank.

5.1. Calibration of DOM electronics and PMT

Calibration of PMT gain and electronic components of a DOM are performed during special calibration runs by software called 'DOMCal' running on the DOM CPU [1]. The basic calibration quantity is the charge recorded for single photoelectrons which are produced by cathode noise and radioactive decays in the glass pressure sphere (in the following referred to as 'cathode noise').

Charge calibration. The charge calibration of the ATWDs starts by converting the ADC counts of each bin to voltages. The ADC-count to voltage relation is determined by varying the ATWD reference voltage and applying a linear fit to the resulting amplitudes yielding slopes and intercepts for each ATWD bin. The 'pedestal pattern', given by the intercepts, is equalized for all bins to a common baseline by the DOM firmware. This common baseline is continuously determined during the data taking runs as described in Section 4.2.3. Since the pedestals of all ADC cells are equalized, only one number for a channel's baseline has to be obtained by averaging for a certain time period, usually for a run of about eight hours duration. The baselines are continuously monitored and the calibration constants are updated when needed.

The ATWD front-end amplifiers are then calibrated in a two-step process. First, the channel with the 16× pre-amplifier is calibrated by injecting single photoelectron-like pulses with the pulser on the mainboard, and comparing the measured pulse charge to the known true pulse charge from the pulser. The other channels are then calibrated consecutively relative to the highest-gain channel using LED light injected into the PMT.

The PMT gain is calibrated using single photoelectron pulses from cathode noise. The charge distribution obtained from these pulses can be described by a Gaussian single-photoelectron peak and an exponential contribution at low charges [11]. The relation between high voltage setting and single-photoelectron peak is determined by varying the bias voltage between 1200 and 1900 V. These voltages are higher than those used for the low gain PMTs in IceTop (750 V and 1250 V for low gain and high-gain PMTs, respectively, Table 4) so that an extrapolation is necessary. The observed single-photoelectron pulses allow the absolute calibration of charges in units of



Figure 17: Distribution of muon track lengths in a tank for a $\cos^{2.3} \theta$ zenith angle distribution.

PE with an estimated precision of 10%. Note, however, that for IceTop the final precision is given by the VEM calibration as described below.

While for in-ice DOMs the discriminator thresholds are calibrated in terms of single photoelectron charges the IceTop discriminator thresholds are set at fixed voltages as described in Section 3.3.3. The voltages can be related to a charge in numbers of PE or VEM for average IceTop pulses as demonstrated in Fig. 13. The charge values are obtained from the calibrated ATWDs.

Timing calibration. The pulse transit times from the PMT to the recording are determined using pulses from the LED on the DOM mainboard (Fig. 10 b).

The master clock of the IceCube experiment is a GPS clock in the IceCube Lab. The local 20 MHz oscillator of each DOM is compared to this master clock in a procedure called "reciprocal active pulsing calibration" (RAPcal) [10] at a frequency of 1 Hz. Each signal from any IceCube detector component gets a timestamp with the same time reference expressed in UTC times. From experimental data the precision of the time calibration was found to be better than 3 ns [1].

The RAPCAL process involves analog measurements on pulses exchanged between the IceCube Lab and individual DOMs, and was optimized for the dispersion and attenuation characteristics of in-ice cables. Since the IceTop cables are shorter, electronic filters are used to make the corresponding pulse shapes similar to in-ice pulses.

5.2. Tank calibration using atmospheric muons

The DOM response to the same energy deposition in a tank varies from tank to tank because of different optical properties of the tanks and DOM efficiencies. Therefore, the tank signals are calibrated by normalizing to the signal charge of a vertical muon defining the unit 'Vertical Equivalent Muon' (VEM) as described in Section 3.3.1.

The VEM calibration procedure uses the natural flux of atmospheric muons. Since 2009 a single-muon calibration trigger is run continuously together with the normal data taking, which has several advantages as compared to employing special calibration runs as done before. In particular, there is no detector downtime, and the calibrations can be done more regularly.

5.2.1. Single muon spectra

The average muon energy at the detector level is in the order of 2 - 2.5 GeV. Since muons at these energies are minimum ionizing, they always lose about the same amount of energy for a given track length in the tank. For instance, the mean energy deposit of a minimum ionizing vertical muon in a tank of 90 cm ice (density 0.92 g/cm²) is about 165 MeV. Cherenkov light is emitted either by the muon itself or by the secondary electrons along the muon track. The total number of Cherenkov photons scales with the energy deposit, and thus with the muon track length, in the tank. Therefore, the spectrum of muon signals in a tank is mainly determined by the angular distribution of the muons and the tank geometry. Figure 17 shows a purely geometrical calculation of the track length distribution in a tank, using a realistic muon zenith angular distribution $dN/d\Omega \sim \cos^{2.3} \theta$ [20]. The sharp peak at 90 cm is mainly due to vertical tracks passing through top and bottom of a tank whereas shorter track lengths are entirely due to edge-clippers. Because of the relation between the track length and the number of Cherenkov photons, the experimental muon spectrum (Fig. 18 a) exhibits the same features but they are smeared out due to statistical fluctuations of the Cherenkov photons, the optical properties of the tanks, and the DOM electronics. Furthermore, there is a background under the muon peak mainly from electromagnetic cascade particles from low energy air showers.

5.2.2. Taking calibration data

The calibration of the IceTop tanks is based on signal spectra from events with preferentially only one muon hitting the tank. These spectra are recorded for high-gain DOMs in parallel to the normal data taking by special calibration DOM launches, called VEMCal launches, which require the LC condition between the DOMs of the same station not to be fulfilled (see section 4.2.3). With this anti-coincidence condition, contributions from extensive air showers are largely suppressed and those of single muons are enhanced. The DOM resident firmware pre-scales the discriminator trigger rate by a factor of $8192 (= 2^{13})$ and launches the DOM readout if there is no LC fulfilled, otherwise the following discriminator trigger is taken. This results in a DOM launch rate of about 0.2 Hz per DOM. For each of these VEMCal launches the ATWD waveforms are sent to the IceTop DOMHub (Section 4.3.1). When a hub receives such data it generates a calibration trigger which initiates event building around the hit (Section 4.3.2) and passes the data to the online processing.

VEMCal launches are only enabled for the high-gain DOMs since the gain and discriminator thresholds of the low-gain DOMs do not allow triggering on single muons. For low-gain DOMs the VEM calibration has to be determined from the overlap of charge measurements by each DOM in a tank, as will be explained below.

The VEMCal hits are calibrated and extracted during the online processing at South Pole exactly the same way as the physics data which are (currently) processed in the North. The raw waveform, given in ATWD channel counts, is calibrated using the DOMCal calibration constants (see Section 5.1). Corrections for residual baseline offset and a droop correction are also applied. Finally, the charge, given in PE units, is calculated by summing up all the waveform bins. The extracted VEMCal hits are stored in a special data container and transferred to the North within the normal physics data stream, while the original VEMCal events are only stored on tape (unless they satisfy other filter conditions).

5.2.3. Processing of calibration data

In the North the VEMCal hit data for each individual DOM are collected for one week and analyzed. The analysis involves fitting of two plots : The fit to the muon spectra to extract the muon peak of the high-gain DOMs and a fit of the correlation between high-gain and low-gain charges to cross-calibrate the low-gain DOMs.

The muon spectra are fitted with an empirical formula which includes vertical muons, edge clippers and the electromagnetic background [21]:

$$f(x) = \underbrace{p_0 \left[L(x; p_1, p_2) + \frac{1.85}{p_1} \cdot \frac{1}{\exp\left(\frac{x - p_1}{p_2}\right) + 1} \right]}_{f_u} + \underbrace{p_3 \cdot \exp\left(p_4 \cdot x\right)}_{f_{em}}$$
(7)

The first term in the muon part, L (x; p_1 , p_2), describes through-going muons by a normalized Landau distribution (from ROOT library [22]), the second the edge-clipping muons by a Fermi-like function and the f_{em} term describes the electromagnetic background by a single exponential function with two parameters (p_3 , p_4). The parameter p_0 is the number of muons which are not edge-clipping, p_1 and p_2 are the location and width parameters of the Landau distribution, respectively. The number of edge-clippers is 1.85 p_0 determined from the purely geometrical considerations in Fig. 17; the Fermi function is normalized for positive x values by the factor $1/p_1$. Figure 18 a shows a fit of the function (7) to a charge spectrum.

5.2.4. VEM definition

The unit 1 VEM is defined as the charge value at 95% of the muon peak which is the maximum of the muon contribution in the fit (term f_{μ} in (7)). The measured muon peak position, obtained with a well-defined fitting procedure, is the essential calibration reference. Shifting the VEM definition to 95% of the peak value has been chosen because the measurement of tagged vertical muons resulted in a 5% lower muon peak (see Section 5.3). However, the choice of a 5% shift does not affect physics results because it also appears in simulations. The charge 1 VEM corresponds to about 125 PE for most high-gain DOMs except for those commissioned in 2005 (the tanks are listed in Table 2) which have about 200 PE (Fig. 19). The low-gain DOMs have systematically lower VEM charges of about 105 PE with correspondingly higher values of the 2005 tanks. This systematic shift between



Figure 18: Calibration monitoring plots of the DOMs in tank 61A from April 15, 2012. a) Charge spectrum of the high-gain DOM in Tank 61A. The black curve corresponds to a fit employing the function (7) with the muon (green) and electromagnetic (red) contributions. The maximum of the muon contribution is indicated by the green arrow while the black arrow points to the value of 1 VEM at 95% below the maximum. b) Relative charge difference of the high-gain (HG) and low-gain (LG) DOMs in tank 61A. The red curve is the average charge difference for a given LG charge slice which is fitted by the green curve according to the function (8).



Figure 19: Histogram of the VEM values of all the high-gain and low-gain DOMs from VEM calibration April 15, 2012.

high-gain and low-gain DOMs is not fully understood (but accounted for by the VEM calibration), see discussion in the next Section 5.2.5.

The energy of air showers can only be determined by comparison with simulations. It is therefore essential that the simulation uses the same definition of VEM to quantify the tank response to air showers. Here the merit of the VEM calibration becomes particularly obvious: the details of each tank do not have to be simulated, but one typical tank behavior can be used in the simulation, and the simulation can be approximately applied to all tanks when signals are expressed in units VEM. The 'calibration' of the simulated signals in units of VEM is described in Section 9.3.

5.2.5. Low gain cross-calibration

Since the muon spectra are only recorded and fitted for the high-gain DOMs the low-gain DOMs are crosscalibrated assuming that the two DOMs of a tank record on average proportional light yields per particle. Therefore the signal ratio of the two DOMs should be constant in an overlap region where saturation does not play a role. Figure 18 b shows the relative charge differences between high gain and low gain versus the low-gain charge in units of PE. The mean values of the scatter plot at a given low-gain charge are constant up to about 2200 PE where saturation of the high-gain DOMs starts.

For each tank the average relative charge differences as a function of the low-gain charge are fitted by the expression

$$f(x) = \begin{cases} p_0 & \text{for } x < p_3\\ p_0 + p_1 \log_{10}(1 + p_2 (x - p_3)) & \text{for } x \ge p_3 \end{cases}$$
(8)

with the free parameters p_0 , p_1 , p_2 and p_3 . The cross-over between the two function parts in (8) is at $x = p_3$.

The horizontal line to the left of the cross-over has an offset $p_0 = \Delta r$ from zero corresponding to the relative difference in the efficiencies of both DOMs which are systematically lower for low-gain DOMs as can be seen in Fig. 19. The efficiency includes the quantum efficiency of the PMT as well as the optical efficiencies of the tank which may vary within a tank due to inhomogeneities in absorption and scattering. Furthermore a potential miscalibration of the gain of the low-gain DOMs, likely due to extrapolation of the PMT gain below the single-photon calibration range (Section 5.1), could be the cause for the systematic shift between low-gain and high-gain DOMs. The offset from zero can be converted to a correction factor for the low-gain DOM for the charge of a VEM (or any other fixed signal) according to:

$$Q_{LG}(\text{VEM}) = \frac{1}{1 - \Delta r} Q_{HG}(\text{VEM}).$$
(9)

The left side is the VEM value of the low-gain DOM in units of PE if Q_{HG} (VEM) is expressed in units of PE. The cross-over at $x = p_3$ (in terms of low-gain charge) indicates the beginning of the saturation of the high-gain PMT and determines the switching from high-gain charges to low-gain charges for the waveform measurements. In terms of high-gain charge (in units PE) the cross-over is at

$$Q_{HG}^{cr-ov} = (1 - \Delta r) \, p_3 \,. \tag{10}$$

5.2.6. Monitoring VEM calibration

The VEM calibration as shown in Fig. 18 is done for all DOMs in an automated way. Thus stability checks and long time DOM response changes can be monitored, as depicted in Fig. 20 for the DOMs in tank 61A (for DOM numbering see Fig. 6). The variations of the number of photoelectrons per VEM for all DOMs between two successive calibrations are within 10% with an rms spread of 3%. The increase of the S/B ratio of the muon signal to the electromagnetic background (see definition in Eq. (14)) indicates increasing absorption of the electromagnetic shower component and is related to the increase of the snow height on the tank (Section 6.3). The discriminator trigger rates plotted below the S/B ratio are mostly determined by the electromagnetic shower component and thus decrease with increasing snow.

5.3. Comparison to runs with muon taggers

Muon telescope at the South Pole. A portable, solar-powered muon telescope based on scintillation detectors was employed in order to tag nearly vertical muons for calibration studies. Comparing the tagged muon events to simulations allows verifying the shower generation and the tank simulation. The variation of the tank response as a function of muon position and the tank-to-tank variations have also been investigated [23].

Measurements were done in the polar season 2005/2006 on tanks deployed one year earlier. The muon tagger was placed on top of an IceTop tank, and all hits in both DOMs were recorded for six hours. Figure 21 shows the charge spectra for the two DOMs in tank 39B and superimposed the tagged muon charge spectra with the muon telescope placed above the center of the tank. The muon telescope spectrum peaks about 5% lower than the calibration spectrum because in the latter case the accepted zenith angle range is wider leading to on average longer particle trajectories in the tank. These measurements are the justification behind choosing the definition of one VEM to be 95% of the muon peak, as described in Section 5.2.4.

Zenith dependence of muon flux. The muon flux at the South Pole was measured for five zenith angles, 0° , 15° , 35° , 82.13° and 85.15° , with a scintillator muon telescope incorporating an IceTop prototype tank as the absorber [24]. Besides the flux measurements also the zenith angle dependence of the recorded waveforms were analyzed and compared to simulations.

The comparison of waveforms from vertical and nearly horizontal muons showed two features in the data: (i) The average amplitude of horizontal muons is about two times as big as that of the vertical muons. (ii) The average rise time of the these pulses in the tank is 3 ns longer than that of vertical muon pulses while the decay time of the average waveform is nearly the same as that of the average waveform of vertical muons. Both features were well reproduced by a Geant4 simulation of the tank response.



Figure 20: Calibration history of the high-gain DOM (a) and low-gain DOM (b) in tank A of station 61 (same as in Fig. 18). From top to bottom are displayed: The number of photoelectrons per VEM, the high-gain - low-gain cross-over used in the calibration of low-gain DOMs, the ratio of the muon signal to the electromagnetic background from the fit of the muon spectrum, the discriminator trigger rate (MPE for high-gain DOMs and SPE for low-gain DOMs), the high voltage of the PMT and the temperature measured on the DOM board. The green vertical lines indicate the time of DOMCal runs where in general DOMCal constants can change.



Figure 21: Charge spectra of a high-gain DOM (a) and low-gain DOM (b) in tank 39B collected by the calibration procedure (VEMCal, black lines) compared to the spectrum tagged by the Muon Telescope (filled histogram) [23].

Laboratory calibration measurements. To study the response of an IceTop tank in more detail a laboratory setup has been developed. It comprises an IceTop tank filled with 76 cm of water (not frozen), equipped with two optical modules with analog readout. Tiled plastic scintillator plates, mounted around the tank, are operated in a coincidence mode. They allow the selection of muon tracks passing through the tank at defined angles and positions. For each position and angle, typically 10^4 muons were measured and the mean value and width of the distribution of the number of photoelectrons in each optical module was determined. Since the dimensions (76 cm height vs. 90 cm) and optical properties of the active medium (water vs. ice) were not the same as at South Pole, the aim of the test was (i) to confirm that the results are correctly reproduced by the simulation program 'Tanktop' (described in Section 9.2.1) which is the reference for the IceTop simulation (Section 9.2), and (ii) to gain a better understanding of possible systematic uncertainties of the calibration measurement.

A scan of nearly vertical muons passing through the tank at different *x*-*y*-positions has been performed. In Fig. 22 a top view of the tiled scintillator centered on the IceTop tank is presented. The left plot shows for each segment the mean number of photoelectrons generated by muons which enter through the segment and exit through the corresponding segment below the tank. The signal variations due to different distances of the muon track to the



Figure 22: Top view of the tiled scintillator centered on the IceTop tank. The mean number of photoelectrons generated by nearly vertical muons entering through each segment and measured in an optical module located at (x = -0.285 m, y = 0) is shown. The measurements (left) are compared to simulations with the simulation code 'Tanktop' (right).



Figure 23: Mean measured number of photoelectrons for muons passing with different track length through a water-filled tank. The track lengths are selected under different zenith angles by the scintillator coincidences as described in the text. The same track length can be obtained under different zenith angles. The measurements (blue) are compared to simulations.

optical module, indicated by the small circles, amount to about 40% maximum and are described to within 10% by the simulation shown in the right plot.

Figure 23 shows the mean measured signal strength in units PE for different muon track lengths in the tank. The signal strength scales nearly linearly with the effective track length. Since the track lengths have been selected by the scintillator coincidences each measurement corresponds to a zenith angular range. The average zenith angle for each point is also displayed in the figure. The measurements are well reproduced by the simulation.

Further investigations of the zenith angle dependence of signal sizes are anticipated, since the test tank has not the same geometrical dimensions as the deployed tanks. In addition, a data-based study, e.g. exploiting isolated muons on the periphery of well-reconstructed showers, would allow better control of angle-dependent systematic uncertainties in physics analyses.

5.4. SLC charge calibration

For SLC hits (Section 4.2.2) integrated charges and time stamps are delivered by the DOM firmware. Since the baseline calibration as described in Section 4.2.3 is not available at this point, the baseline correction is applied off-line using the same baseline determination as for HLC hits. Remaining differences to the HLC charges are corrected by comparing the correlation between HLC and SLC pulses when both are available.



Figure 24: The temporal behavior of the South Pole stratosphere from May 2007 to April 2009 is compared to IceTop DOM counting rate and the high energy muon rate in the deep ice. (a) The temperature profiles of the stratosphere at pressure layers from 20 hPa to 100 hPa where the first cosmic ray interactions happen. (b) The IceTop DOM counting rate (black -observed, blue -after barometric correction) and the surface pressure (orange). (c) The IceCube muon trigger rate and the calculated effective temperature (red) [25].

6. Environmental conditions at South Pole

6.1. Overview

Since cosmic ray measurements with IceCube use the Earth's atmosphere as a converter medium, variations of the atmosphere will affect air shower measurements. The shower development is mainly determined by the overburden, X(h) = p(h)/g, at a height *h* and a pressure p(h) (*g* is the Earth acceleration, here assumed to be constant). Pressure measurements are available at the South Pole from balloon flights. In addition to the total overburden, as given by the ground pressure (at South Pole the average is about 680 hPa), also the density profile of the atmosphere influences the air showers. A change in density changes the balance between strong interactions and weak decays of mesons and thus the rate of lepton production. The high-energy muon rate observed in the deep ice is mainly influenced by the density (or temperature for a given pressure) profile in the stratosphere.

The South Pole atmosphere undergoes a pronounced annual cycle. In the winter months, April to September, when the sun is below the horizon, the air is much colder than in summer, with surface temperatures ranging from -20 °C down to -70 °C. Therefore, the winter atmosphere is much denser than in summer. Figure 24 shows the seasonal variation of the density in the upper atmosphere and the correlation with IceTop DOM rates and inice muon rates [25]. The high-energy muon rate in the deep ice follows closely these density changes with an approximate $\pm(8-9)\%$ variation (Fig. 24c), while the variation of the IceTop DOM rates is about $\pm5\%$ (Fig. 24b).

In contrast to the strong annual variation of the density profile, there is no cyclic variation of ground pressure. Instead changes happen on a much shorter time scale, as seen in Fig. 24b. While the high-energy muons rates are not influenced by the ground pressure changes, there is a strong correlation between pressure and IceTop DOM rates. An increase in air pressure causes a stronger attenuation of air showers leading, for a given shower, to a decrease of the observed shower size in IceTop. The related shift of primary energy thresholds makes the IceTop rates anti-correlated to air pressure. The pressure effect can empirically be corrected for (blue curve in Fig. 24 b), which is useful for rate monitoring. For air shower reconstruction it is necessary to find a correction for the shower size S, the observed shower signal at a reference radius, which is the most important observable of an air shower (the determination of S is described in Section 8).

Another effect which influences the shower development is the snow accumulating on the tanks. While the atmosphere affects the overall profile of an entire shower, snow accumulation affects individual tanks depending on how deeply they are buried.



Figure 25: Distribution of snow heights on top of IceTop tanks measured in February, 2012, and interpolated between tanks. The labels point at buildings which affect snow drifting and may give rise to special snow management.

6.2. Atmosphere

6.2.1. Barometric effect on rates

Empirically, the change in the IceTop rates with the surface pressure P as shown in Fig. 24 depends almost linearly on the absolute pressure change:

$$\frac{dN}{N} = \beta \, dP \quad \Rightarrow \quad N = N_0 \, \exp(\beta \, (P - P_0)). \tag{11}$$

The barometric coefficients β , introduced by this relation, are regularly determined for the various trigger and filter rates by fitting data at different pressures. The fitted β values range from about 0.6 to 0.7 %/hPa for different rates. For monitoring purposes a rate N_i^{meas} measured at a particular pressure P_i is normalized to a rate at the reference pressure P_0 according to

$$N_i^{corr} = N_i^{meas} \exp(-\beta \left(P_i - P_0\right)).$$
⁽¹²⁾

After correcting the DOM rates for surface pressure variations (blue curve in Fig. 24b) seasonal effects are clearly visible in the DOM rates, although not as strong as for the in-ice muon rates in Fig. 24c.

6.2.2. Effect of the atmosphere on energy measurements

At different pressures air showers are differently absorbed in the atmosphere so that an effect on the measured shower size is expected. Corrections are derived by comparing measured shower size spectra from periods with different surface pressure and finding the correction which makes the spectra coincide. Experimentally it is found that the shower size parameter *S* changes with the surface pressure *P* with respect to a reference pressure P_0 as

$$S(P) = S(P_0) \exp\left(-\frac{P - P_0}{\lambda_P(S, \theta)}\right) \quad \text{or} \quad \frac{d\ln S(P)}{dP} = -\frac{1}{\lambda_P(S, \theta)}.$$
(13)

The absorption parameter $\lambda_P(S, \theta)$ is studied as a function of the shower size S and the zenith angle θ . For vertical showers above 1 PeV one finds $1/\lambda_p \approx 0.25 \times 10^{-5} \text{ Pa}^{-1}$ or, in terms of overburden, $1/\lambda_p \approx 0.25 \times 10^{-4} \text{ cm}^2/\text{g}$ [26].

The seasonal effect on the shower size measurements was estimated to be $\Delta \log_{10} S \approx \pm 0.01$ for the two extreme atmosphere densities in winter and summer [26]. Corrections of atmospheric effects on energy determination from reconstructed showers are discussed in Section 8.3.

6.3. Snow on tanks

While precipitation at South Pole only amounts to about 2 cm per year, there is a significant snow accumulation due to drifting. The snow height on top of the tanks increases on average by 20 cm every year with a density of 0.35 to 0.4 g/cm^3 depending on depth. However, accumulation strongly depends on surrounding terrain and buildings. At South Pole the wind has a predominant direction and a relatively constant average speed of about 5 m/s.This



Figure 26: Snow heights measured in Feb. 2010, Dec. 2010 and Feb. 2011: a) The ratio S_{μ}/B_{EM} versus the measured snow height. b) Comparison of measured and extrapolated snow heights. The extrapolation is done by using the signal-over-background ratio of the VEM calibration with formula (7).

causes an increased accumulation of snow leeward of buildings and slopes. Figure 25 shows the distribution of snow on top of IceTop stations in February 2012.

Snow depths on the tanks can directly be measured only during the Antarctic summer. However, the development of snow heights during the whole year is important for the analyses of the data. For the determination of the snow heights between direct measurements the calibration spectra as shown in Fig. 18 are employed. The effect of snow is mainly an absorption of the electromagnetic shower component while the muon spectrum remains nearly unaffected by growing snow. Therefore, the ratio of the number of muons around the muon peak in Fig. 18 to the underlying mainly electromagnetic background depends on the snow height. This ratio is defined by integrating over the muon part f_{μ} and the electromagnetic part f_{em} of the calibration fitting function (7):

$$S_{\mu}/B_{EM} = \int_{S_{min}}^{S_{max}} f_{\mu} \, dS \, \left| \int_{S_{min}}^{S_{max}} f_{em} \, dS \right| \tag{14}$$

Suitable integration boundaries have been found to be $S_{min} = 0.3$ VEM and $S_{max} = 2.0$ VEM.

Figure 26 a shows the correlation between measured snow heights and this ratio. The points can be well fitted by the function:

$$h_s = a_s \times (\ln(S_\mu/B_{em}) + b_s) \tag{15}$$

typical values of the parameters are $a_s \approx 1.37 m$ and $b_s \approx 1.75$.

Formula (15) is used to estimate snow heights in between direct measurements. For a correct extrapolation the slope a_s is fixed to the average of all snow heights but the intercept b_s is determined for each tank separately. A comparison of snow heights determined in this way with direct measurements is shown in Fig. 26 b. The fitted snow heights are used to account for the snow effects in the reconstruction of shower parameters (Section 8.3).

7. Signal processing and data preparation

Figure 11 shows typical waveforms measured in IceTop. For air shower reconstruction, these waveforms are processed by a series of software modules to perform three steps: calibration, extraction of total pulse charge and time, and pulse cleaning. Only the unsaturated ATWD channel with the highest gain is used in the data analysis.

First, the ATWD baseline (common for all bins, see Section 5.1) is subtracted from the digitized ATWD waveforms. The waveforms are then converted from ADC counts into voltage taking into account the gain of the ATWD amplifier and the calibration of each individual ATWD bin. In case of SLC hits, where no waveform is available, the average ADC calibration of bins 5 to 60 is used. The start time of the waveform is corrected for the transit time of the pulse through the PMT and the electronics (Section 5.1). Finally, all waveforms are corrected for the effects of droop caused by the transformer that couples the PMT output to the recording electronics (Section 3.5). The correction uses the inversion of formula (3) in Section 3.5 in a discretized way. This is done by correcting the voltage in each time bin for the droop in the same bin and for the summed voltage responses from all preceding bins. When a readout window contains consecutive hits from the same DOM, the waveform of the last hit is used



Figure 27: Example of the IceTop air shower reconstruction. a) Lateral fit to signals from 25 triggered stations with a reconstructed shower size $S_{125} = (65.1 \pm 2.8)$ VEM. b) Time residuals with respect to a plane perpendicular to the shower direction given by Eq. (21). "Upstream" and "downstream" refer to tanks being hit before and after the shower core reaches the ground.

to correct additionally for the residual droop in the follow-on hit. Droop correction cannot be applied to SLC hits but is on average accounted for by the SLC calibration (Section 5.4).

The total HLC pulse charge is then extracted by integrating the calibrated waveform and converting it into units of PE and VEM to be used in reconstruction. The pulse time is defined by the crossing of the leading edge slope of the first pulse with the baseline. The slope is taken between 10 to 90% of the leading edge (see Fig. 11 a). All times are expressed in UTC time.

For each tank, charge and time of the pulse in the high-gain DOM are used, unless the charge surpasses a saturation threshold determined during the VEM calibration (expression (10) in Section 5.2.5). In that case, the charge is used from the low-gain DOM. The pulse time is always based on the high-gain DOM. If there is no pulse in a low-gain DOM within ± 40 ns of a saturated high-gain pulse, the pulse is marked as saturated to be treated accordingly by reconstruction algorithms.

In addition, for air shower reconstruction, events are cleaned by requiring the following conditions:

• A station is discarded if the following condition on the signal times in the two tanks A and B is not met:

$$|t_A - t_B| < \frac{|x_A - x_B|}{c} + 200 \,\mathrm{ns},$$
 (16)

where t_A and t_B are the signal times in the two tanks located at x_A and x_B .

• Stations are grouped in clusters which in principle could belong to the same shower, such that any pair of stations *i* and *j* in the cluster fulfills the condition

$$|t_i - t_j| < \frac{|\mathbf{x}_i - \mathbf{x}_j|}{c} + 200 \,\mathrm{ns.}$$
 (17)

The station position x_i is the center of the line connecting its two tanks, and t_i is the average time of both tank signals. In each event, only the largest cluster of stations is kept.

This selection is done in order to remove obviously unrelated pulses from events. In the IT26 analysis [27] it only affected about 4% of events, where on average 2.3 tanks were removed.

8. Air Shower Reconstruction

Properties of a primary particle are inferred from the air shower parameters reconstructed from the IceTop signals. The reconstructed parameters include the shower core position and direction, and the shower size. The latter is a measure of primary energy and is defined as the signal S_{ref} measured at a certain distance R_{ref} from the shower axis. These properties are reconstructed by fitting the measured charges with a lateral distribution function and the signal times with a function describing the geometric shape of the shower front (Fig. 27). Currently only HLC hits are used in the reconstruction.

8.1. Time and charge distribution of air shower signals

Lateral charge distribution. IceTop tanks are not only sensitive to the number of charged particles, but also detect photons via electron-positron pair production. That is different from a scintillator array, which is relatively insensitive to the photonic part of the signal resulting in a lateral charge distribution which can be described by the NKG function [28, 29]. Therefore, for IceTop an empirical lateral distribution function was derived from simulations [30, 31].

The charge expectation value in an IceTop tank at distance R from the shower axis is described by:

$$S(R) = S_{\text{ref}} \cdot \left(\frac{R}{R_{\text{ref}}}\right)^{-\beta - \kappa \log_{10}(R/R_{\text{ref}})}.$$
(18)

which is a second order polynomial in the logarithm of *R*:

$$\log_{10} S(R) = \log_{10} S_{\rm ref} - \beta \, \log_{10} \left(\frac{R}{R_{\rm ref}} \right) - \kappa \, \log_{10}^2 \left(\frac{R}{R_{\rm ref}} \right). \tag{19}$$

This function behaves unphysically at small distances to the shower axis ($R \leq 1$ m). However, as described in the next subsection, all signals that are closer than a minimal distance to the core, are excluded from the fit. The free parameters of the function, in addition to the shower size, S_{ref} , are β and κ , corresponding to the slope and curvature in the logarithmic representation at $R = R_{ref}$. Simulation studies suggest that κ can be fixed without a significant impact on the fit result. The current default is $\kappa = 0.303$.

At medium energies, the distance from the shower axis corresponding to the average logarithm $\langle \log_{10} R/m \rangle$ of signals participating in the fit is about 125 m. The current standard fit uses a constant $R_{ref} = 125$ m, which was found to minimize the correlation between the parameters S_{ref} and β on average for the IT26 configuration. The shower size parameter is thus referred to as S_{125} . An optimization of this parameter for the full detector is under study.

Time distribution. The arrival times of the signals map out the shower front. The expected signal time of a tank at the position x is thus parametrized as

$$t(\mathbf{x}) = t_0 + \frac{1}{c}(\mathbf{x} - \mathbf{x}_c) \cdot \mathbf{n} + \Delta t(R).$$
⁽²⁰⁾

Here, t_0 is the time the shower core reaches the ground, \mathbf{x}_c is the position of the shower core on the ground and \mathbf{n} is the unit vector in the direction of movement of the shower. For fixing the core position, 'ground' is defined as the \sqrt{S} -weighted average of participating tank altitudes, which varies by about 6 m over the whole array. The term $\Delta t(R)$ describes the shape of the curved shower front as a function of distance R to the shower axis and is the time residual with respect to a plane perpendicular to the shower axis which contains \mathbf{x}_c . A plane through \mathbf{x}_c is described by $\Delta t(R) = 0$, used as start value for the fit, see Section 8.3. Studies on experimental data showed that the shower front can be described by the sum of a parabola and a Gaussian function, both symmetric around the shower axis [31]:

$$\Delta t(R) = a R^2 + b \left(1 - \exp\left(-\frac{R^2}{2\sigma^2}\right) \right), \tag{21}$$

with the constants

$$a = 4.823 \, 10^{-4} \, \text{ns/m}^2$$
, $b = 19.41 \, \text{ns}$, $\sigma = 83.5 \, \text{m}$

The energy, zenith angle and mass dependence of these parameters is under study and presents an opportunity for further improvement of the reconstruction quality. Furthermore, inclined showers will not be symmetric with respect to the shower core, which may require a modification of the signal time and charge distribution.

Function (20) is fitted to the measured signal times with five free parameters: two for the core position, two for the shower direction and one for the reference time t_0 . Hence, the complete air shower reconstruction has the following parameters: position of the shower core (x_c, y_c) , shower direction θ and ϕ , shower size S_{125} , slope parameter β , and time at ground t_0 .

8.2. Shower fluctuations

Fluctuations of signal size and arrival time for each tank have to be included in the fit procedure. The signal size fluctuations are described by a normal distribution of $\log_{10} S_i$ around the fitted expectation value $\log_{10} S_i^{\text{fit}}$, with standard deviations $\sigma_{\log_{10} S}$ depending on the signal charge. The charge dependence of $\sigma_{\log_{10} S}$ has been determined

experimentally from the local shower fluctuations between the two tanks of a station and are reasonably well reproduced by simulation [32]. For the shower reconstruction a functional description is used:

$$\log_{10} \left(\sigma_{\log_{10} S} \right) = \begin{cases} 0.283 - 0.078 \log_{10} \left(S / \text{VEM} \right) & \log_{10} \left(S / \text{VEM} \right) < 0.340 \\ -0.373 - 0.658 \log_{10} \left(S / \text{VEM} \right) + 0.158 \log_{10}^2 \left(S / \text{VEM} \right) & 0.340 \le \log_{10} \left(S / \text{VEM} \right) < 2.077 \\ 0.0881 & 2.077 \le \log_{10} \left(S / \text{VEM} \right) \end{cases}$$

The arrival time fluctuations are described by a normal distribution with a standard deviation $\sigma_t(R_i)$,

$$\sigma_t(R_i) = 2.92 \,\mathrm{ns} + 3.77 \cdot 10^{-4} \,\mathrm{ns} \cdot (R_i/\mathrm{m})^2 \,, \tag{23}$$

(22)

depending on the distance R_i of tank *i* to the shower axis as found in experimental data [32]. No energy dependence of σ_t has been included so far.

8.3. Likelihood fit

Likelihood function. The functions (19), (20) and (21) describing the expectations for the charge and time of air shower signals are fitted to the measured data using the maximum likelihood method (Fig. 27). In addition to the signal charges and times, the likelihood function also takes into account stations that did not trigger so that the full log-likelihood function consists of three terms:

$$\mathcal{L} = \mathcal{L}_q + \mathcal{L}_0 + \mathcal{L}_t. \tag{24}$$

The first term,

$$\mathcal{L}_{q} = -\sum_{i} \frac{(\log_{10} S_{i} - \log_{10} S_{i}^{\text{fit}})^{2}}{2 \sigma_{\log_{10} S}^{2} (S_{i}^{\text{fit}})} - \sum_{i} \ln(\sigma_{\log_{10} S} (S_{i}^{\text{fit}})),$$
(25)

describes the probability of measuring the charges S_i if the fit expectation value at the position of the tank is S_i^{fit} as given by the lateral distribution function (18). The standard deviation $\sigma_{\log_{10} S}$ is given by (22). The sum runs over all tanks that have triggered. The second sum in \mathcal{L}_q accounts for the proper normalization of the signal likelihood and has to be added because the standard deviations depend on the fitted signals.

The next term of the log-likelihood function (24),

$$\mathcal{L}_0 = \sum_j \ln\left(1 - (P_j^{\text{hit}})^2\right),\tag{26}$$

accounts for all stations *j* that did not trigger. The probability that one tank in station *j* delivers a signal at a given charge expectation value is

$$P_{j}^{\text{hit}} = \frac{1}{\sqrt{2\pi}\sigma_{\log_{10}S}(S_{j}^{\text{fit}})} \cdot \int_{\log_{10}S_{j}^{\text{thr}}}^{\infty} \exp\left(-\frac{(\log_{10}S_{j} - \log_{10}S_{j}^{\text{fit}})^{2}}{2\sigma_{q}^{2}(S_{j}^{\text{fit}})}\right) \mathrm{d}\log_{10}S_{j}.$$
 (27)

The lower integration limit is defined through the charge threshold S_i^{thr} for the tank signal. The charge expectation value, S_i^{fit} , is evaluated for the center of a line joining the centres of the two tanks. Since the two tanks of one station operate in coincidence and tanks with unusually missing partners, as described in Section 4.2.4, are removed by the cleaning described in Section 7, there are no single untriggered tanks. There is a natural correlation in the signal expectation values of two nearby tanks because they have a similar value of the lateral distribution function. However, in Eq. (26) the no-hit probability of the are assumed to be independent because the fluctuations about the average expectation value, S_j^{fit} , of both tanks are uncorrelated. The third term of (24), \mathcal{L}_t , describes the probability for the measured set of signal times,

$$\mathcal{L}_t = -\sum_i \frac{(t_i - t_i^{\text{fit}})^2}{2\sigma_t^2(R_i)} - \sum_i \ln(\sigma_t(R_i)),$$
(28)

where the index *i* runs over all tanks, t_i is the measured signal time of tank *i*, $t_i^{\text{fit}} = t(\mathbf{x}_i)$ is the fitted expectation value according to function (20), and σ_t is defined in (23).



Figure 28: Reconstruction quality of showers in two zenith angle ranges for proton showers with 5 or more stations triggered in the IT73 configuration. Shown are the 68% resolutions as a function of primary energy. a) Distance between true and reconstructed shower core. b) Angle between true and reconstructed shower direction.

Fit procedure. The likelihood fit is seeded with first-guess calculations for the core and the direction of the shower. As a first estimate of the core position the centre-of-gravity of the positions x_i of the *m* tanks with the highest signals weighted with the square root of the charges is calculated:

$$\boldsymbol{x}_{\text{COG}} = \frac{\sum_{i} \sqrt{S_{i}} \boldsymbol{x}_{i}}{\sum_{i} \sqrt{S_{i}}}.$$
(29)

The weight $\sqrt{S_i}$ is chosen based on a study of the achievable fit accuracy. The restriction to the *m* tanks with the highest signals (current default is m = 7) yields a better estimate, in particular near the boundaries. The starting value for the direction is obtained by fitting to the signal times of the *m* tanks with the highest signals a plane, that means setting $\Delta t(R) = 0$ in the function (20).

The likelihood minimization is then done in several iterations to improve the stability of the fit. At first the shower direction is fixed and only the lateral fit of the charges is iterated with the free parameters S_{125} , β , and core position. After each iteration those tanks are removed from the fit that are closer than a minimal distance R_{min} to the shower axis. Iteration is stopped when no more tanks are removed from the fit. The parameter R_{min} is optimized to improved core resolutions by mitigating the effect of saturated pulses and of biases in the determination of core positions¹⁰. Then, a third iteration step is done in which the shower front curvature is included and the shower direction is varied. Finally the direction is fixed again and S_{125} , β , and core position are allowed to vary for a fine tuning.

The fit performance is demonstrated in Fig. 28 where the energy dependence of the core and angular resolutions for the full likelihood fit are displayed. The performance for the energy reconstruction is discussed in Section 10.2.

Small Shower reconstruction. If only 3 or 4 stations are included in the fit, the last two iteration steps are dropped because the small number of stations with small separation between the tanks of a station does not offer sufficient redundancy. The shower direction and the t_0 parameter are taken from the first guess fit assuming a plane shower front.

Snow correction in reconstruction. During the fit procedure the absorption of shower particles due to snow accumulation on the tanks and the snow around the tanks is corrected for. The correction is applied to the fit values in each iteration rather than to the measured charges. In this way the 'no-hits' probability for signals due to the thresholds can be correctly computed using Eq. (27). For the correction an exponential absorption model is assumed:

$$\hat{S}_{i}^{corr} = \hat{S}_{i} \exp\left(-\frac{h_{s}^{i}}{\lambda_{s} \cos \theta}\right).$$
(30)

Here \hat{S}_i is the charge expectation for the tank *i* given by the lateral distribution function and \hat{S}_i^{corr} the corresponding snow corrected value; h_s^i is the snow height above tank *i*, θ is the zenith angle of the shower axis, and λ_s is an effective absorption length in snow. We are studying if an effective absorption describes correctly the interplay between absorption and regeneration in a shower and the different absorption behavior of different particles or if

¹⁰In the IT26 analysis [27] $R_{min} = 11$ m was used.



Figure 29: Shower size spectrum measured with the IT73 configuration in the zenith angular range $\cos \theta > 0.8$ in an area of $5.2 \times 10^5 \text{ m}^2$. The spectrum contains three separate data sets with 3 and 4 stations, 5, 6 and 7 stations and 8 or more stations.

a more complicated description is required. From studies of data and simulations this absorption length has been found to lie between 2 m and 4 m. Dependences of snow attenuation on energy, signal size, zenith angle, and position with respect to the core are still under study.

Atmosphere corrections of shower size. The shower size S_{125} is corrected for surface pressure dependence by inverting formula (13). For the analysis of the IT26 energy spectrum [26] the following corrections were found (replacing in (13) $d \ln S = \ln 10 \log_{10} S$ and the pressure P by the overburden X = P/g):

$$0^{\circ} \leq \theta < 30^{\circ}: \quad \frac{d \log_{10} S_{125}}{dX} = (-1.2 \pm 0.9) \cdot 10^{-4} \frac{1}{g/cm^2},$$

$$30^{\circ} \leq \theta < 40^{\circ}: \quad \frac{d \log_{10} S_{125}}{dX} = (-9.0 \pm 1.5) \cdot 10^{-4} \frac{1}{g/cm^2},$$

$$40^{\circ} \leq \theta < 46^{\circ}: \quad \frac{d \log_{10} S_{125}}{dX} = (-14.3 \pm 2.9) \cdot 10^{-4} \frac{1}{g/cm^2}.$$

(31)

The dependence on the atmospheric density profile has not yet been explicitly corrected for. For data of a whole year the effect should cancel out. For more restricted time periods different atmospheres are studied in simulations (see Section 9.1).

8.4. Shower size spectra and energy determination

A shower size spectrum reconstructed from data taken with the IT73 configuration in 2010/11 is shown in Fig. 29. The zenith angular range is $\cos \theta > 0.8$ and the shower core is required to lie within a polygonal fiducial area with a size of 5.2×10^5 m² excluding all border stations. The spectrum contains three separate data sets with 3 and 4 stations, 5, 6 and 7 stations and 8 or more stations, depicting the different energy ranges.

The relation of the shower size parameter to the cosmic ray energy depends in a non-trivial way on the primary mass composition and the zenith angle and has to be evaluated using simulations (Section 9). A determination of the cosmic ray energy spectrum requires the knowledge of the mass composition. Measurements with IceTop alone can exploit the zenith angular dependence of the shower size on the mass [27]. A better handle on the mass composition is achieved by measuring the air shower on the surface, which is dominated by the electromagnetic component, in coincidence with the high-energy muons in IceCube. The energy spectrum and mass composition have then to be extracted by multi-variate algorithms, for example a Neural Network [33, 34] or two-dimensional unfolding [35].

To get a rough idea of the corresponding energies of the shower size spectrum in Fig. 29, one may use above $S_{125} \approx 1$ VEM the approximate relation $E/S_{125} \approx 1$ PeV/VEM for proton primaries at small zenith angles.

9. Simulation of air showers and the IceTop detector

The relation between the measured signals and the energy of the primary particle, as well as detection efficiency and energy resolution of the IceTop detector are obtained from simulations of air showers generated by the CORSIKA program [36, 37] and then traced through the detector by a Geant4-based [38, 39] program.

9.1. Air shower simulation

Air showers in the atmosphere are simulated using the code CORSIKA. The hadronic component of the air showers is simulated using the models SIBYLL2.1 [40, 41] or QGSJET-II [42, 43] for high energy interactions and FLUKA 2008.3 [44, 45] for low energy interactions (the current default is < 80 GeV). The electromagnetic component is simulated using EGS4 [46]. 'Thinning' (reduction of the number of traced particles [47]) can optionally be applied.

For the South Pole, CORSIKA provides six different pre-defined atmosphere profiles [37]. Four of these atmospheres (CORSIKA labels 11 to 14) are based on the MSIS-90-E model [48], one for each austral season. The two others, modeled by P.Lipari, are alternatives for austral summer and winter. There is also the option to define custom profiles, for example using actual measurements of the South Pole atmosphere by satellites and balloons.

9.2. Detector simulation

The output of the CORSIKA program which includes the shower particle types, positions and momenta at the observation level of 2835 m, are injected into the IceTop detector simulation which uses the Geant4 package. The simulation starts with the generation of the amount of light produced by the shower particles in the tanks followed by the simulation of the PMT, the DOM electronics and the trigger chain.

To make the Geant4-based detector simulation more efficient, optical tank properties are studied in a standalone, single tank simulation (Tanktop). Some of these results are used in a parametrized form for the full detector simulation.

9.2.1. Study of optical tank properties with Tanktop

A standalone program, Tanktop, has been developed allowing detailed studies of the optical properties of the tanks and DOMs. The optical properties are described by wavelength-dependent functions for PMT efficiencies, reflectivities of the tank walls and the ice-Perlite or ice-air boundaries and the absorption in the ice. Reflectivities of the tank wall liners (Fig. 4) and of the perlite filling on top of the ice have been measured under laboratory conditions and have to be tuned for the simulation to obtain agreement with observed signal characteristics. The tuning was done by scaling the reflectivities by a power law $R'(\lambda) = R(\lambda)^p$, with *p* typically about 0.38 (that means that the tuning requires higher reflectivities than measured in the laboratory). The absorption length in ice was taken to be 100 m from measurements in the deep-ice (this value is not critical). For the wavelength dependent PMT efficiencies averages of sample measurements are used [11]. Another study concerned the implementation of the snow cover of the tanks into the simulation.

The simulation studies confirm that an effective, average description of detector properties is sufficient if one employs the VEM concept to calibrate data to a standard signal (Section 5.2). The essential conclusions from the Tanktop studies for the full detector simulation are:

- an effective exponential decay time of pulses due to reflectivities and absorption;
- verification that the number of generated Cherenkov photons in the wavelength interval 300 nm to 650 nm is proportional to the number of detected photoelectrons;
- verification that the latter relation is within the assumptions of the simulation independent of the direction and location of the simulated particle's trajectory in the tank;
- verification that the effect of snow can be included by starting the shower particle tracking just outside the snow volume.

Table 9: Table of materials and their properties used in the Geant4 simulation. The column 'composition' the fractions are given in parentheses.

Material	Composition	Density [g/cm ³]	Conditions/Properties
air	N(0.755267),	0.001205	temperature $T = 243.15$ K,
	0(0.231781),		pressure $p = 67 \mathrm{kPa}$,
	Ar(0.012827),		ionisation potential
	C(0.000124)		$U_I = 85.7 \mathrm{eV}$
snow	H ₂ O	0.38	
ice	H ₂ O	0.92	index of refraction $n = 1.31$,
			optical photon energies:
			$E_{opt} = [1.91 \text{ eV}, 4.13 \text{ eV}]$
plastic	H ₂ CO	1.425	substitute for the tank materials
Perlite	N(0.7018),O(0.2491),	0.1598	mixture of 92.92 % air
	H ₂ O(0.0021), Ar(0.0119),		and 7.08 % Perlite
	Si(0.0240),Al(0.0051),		
	K(0.0025),Na(0.0024),		
	traces of C, Fe, Ca, Mg, Mn		
glass	SiO ₂	2.254	DOM pressure sphere
DOM	SiO ₂	0.2	effective DOM material
			with average density

9.2.2. Tank simulation

The Cherenkov emission inside the tanks is simulated using Geant4. All structures of the tank, the surrounding snow, including individual snow heights on top of each tank, as well as the air above the snow are modeled realistically (see Tables 3 and 9 [49]). The snow heights used in the simulation are implemented for the specific data period as explained in Section 6.3. In order to save computing time, Cherenkov photons are not tracked; only the number of photons emitted in the wavelength interval 300 nm to 650 nm is recorded. This procedure was studied using the stand-alone program Tanktop, described above, which includes Cherenkov photon tracking until photons reach the PMT. The propagation of Cherenkov photons is modeled by distributing the arrival times according to an exponential distribution, which was tuned with Tanktop by varying the optical tank parameters such that simulated waveform decay times match those observed in experimental data (Table 6). In the simulation the number of photoelectrons per VEM is fixed and the same for all tanks (Section 9.3); the corresponding number of photoelectrons are simulated with the exponential time distribution of the real tanks. While the photoelectrons are simulated with the exponential time distribution of the pulse decay (Table 6), a realistic leading edge slope results from the simulation of transport through the electronics (Section 9.2.3).

To further reduce the required computing time, particles whose trajectories miss a tank's volume by more than 30 cm are excluded from the detector simulation. Additionally, when the total charge created inside a tank is larger than 2000 VEM, further particles are not injected into the Geant4 simulation, but only counted. The final number of photoelectrons at the PMT is then scaled accordingly. This significantly reduces the time needed to simulate tank signals very close to the core.

9.2.3. PMT and DOM simulation

In the next step the generated photoelectrons are injected into a simulation of the PMT followed by the analog and digital electronics of the DOM. To simulate the photomultipliers, single photoelectrons with Gaussian shaped waveforms ($\sigma = 3.2$ ns) and a charge randomly chosen from the measured single photoelectron spectrum [11] are superimposed. Pulse saturation is simulated depending on the instantaneous current according to the gain dependent functions depicted in Fig. 30 (PMT saturation is discussed in Section 3.4).

In the DOM simulation, the pulse shaping due to the analog front-end electronics is applied to the output of the PMT simulation. This includes the simulation of the droop effect using a discretized form of formula (3) (Section 3.5) and the individual shaping of the signal paths to the ATWD and the discriminators using formula (1) (Section 3.3). Next, the discriminators are simulated and the local coincidence conditions are evaluated. Finally, the digitization of the waveforms by the ATWDs and fADCs and the array trigger (IceTopSMT, Section 4.3.3) are simulated.

Simulated data have the same format as the experimental data and are reconstructed in the same way, as



Figure 30: Saturation curves as used for the IceTop simulation for the high-gain (red) and low-gain (blue) PMTs. The the curves are labeled with the corresponding gain value. The saturated current is plotted versus the ideal, unsaturated current. The saturation currents given in Table 6 are the unsaturated currents for which the output current is 50% lower.

described in the previous section.

9.3. Definition of VEM units in simulation

The VEM 'calibration' of the tank simulation, that is the transfer of the experimental VEM definition to the simulation, is achieved by relating the number of Cherenkov photons generated by any particle in the tank to a value in units VEM. The relation of 1 VEM to the number of photoelectrons is taken from the calibration data of a typical tank (for each tank type), see Section 5. However, the specific value of PE per VEM is not important for the absolute calibration because it cancels out, but it has an influence on the fluctuations.

The value 1 VEM is assigned to a certain number of Cherenkov photons such that the shower simulation yields a charge spectrum with the correct muon peak position. This assignment is done in an iterative way along the following steps:

- Simulate CORSIKA air showers in an energy range in which showers contribute to the VEMCal spectrum with single muons (empirically: 3 GeV to 10 TeV for an $E^{-2.7}$ spectrum and zenith angles up to 60° [50]) and simulate the number of Cherenkov photons emitted in the tank using Geant4.
- Assign the signal charge of 1 VEM to a certain number of Cherenkov photons (as a start value the Cherenkov photons which a vertical muon of some energy produces).
- Take the relation of PE per VEM from data of a typical tank.
- Simulate the electronic chain for the photoelectrons.
- Apply the same calibration trigger condition as in data (VEMCal launch, Section 4).
- Generate the single muon spectra as in the VEM calibration and fit with the same function as in data formula (7) in Section 5.2).
- Compare the position of the muon peak in VEM units to the data of a tank (if VEM units are used it can be any tank).
- If the simulated peak is not at 1/0.95 VEM tune the VEM assignment to the number of Cherenkov photons (for the factor 0.95 see Section 5.2.4).

The current implementation of the VEM unit into the simulation is: 1 VEM corresponds to 32 235 Cherenkov photons in the wavelength range 300 to 650 nm. This is in Geant4 the yield of a vertical muon with a kinetic energy of about 0.8 GeV.

9.4. Simulation production

9.4.1. Properties of datasets

For cosmic ray analysis, output files from CORSIKA simulation are available for showers initiated by different primaries (proton, iron, helium, oxygen, and silicon), energies between 10^4 and 10^9 GeV and zenith angles of the incoming primaries between 0 and 65°. Simulations up to 40° are used to analyze IceTop events in coincidence with the in-ice detector. Above 40°, simulations are employed for studies of surface events and events with core locations outside the IceTop array area that can still trigger the in-ice detector (for example veto studies, see Section 10.5). In addition, for energies between 10^7 GeV and $10^{9.5}$ GeV, thinned showers (see below) are simulated. The overlap with the energy range of non-thinned showers is used to estimate the systematics induced by thinning.

The simulation datasets are grouped in energy bins of size $0.1 \times \log_{10}(E/\text{GeV})$ with an E^{-1} spectrum, independently normalized in each bin. For comparison with data, the simulations can be weighted to yield a realistic spectrum. The direction zenith and azimuth angles are generated to yield an isotropic distribution (uniformly distributed in $\cos \theta$, for sampling on a horizontal detector surface the distribution is proportional to $\cos \theta d \cos \theta$).

Most of the events are simulated for one specific atmosphere (currently model 12 [37] which is the South Pole atmosphere of July 1, 1997). Smaller data sets generated with other atmospheres are available for studies of atmospheric effects and their corrections. The default interaction models used are SYBILL at higher energies and FLUKA in the lower energy range. Additional datasets generated with the QGSJET model instead of SYBILL are produced for comparison. Compositions according to various models, such as a two-component model [51] or the polygonato model [52, 53], are produced by adding different mass components with an appropriate spectral weighting.

9.4.2. Re-Sampling

Since shower generation is CPU intensive, the same showers are sampled several times before being passed to the detector simulation. Showers are sampled inside a circle with an adaptable, energy dependent radius around the center of the IceTop array. The number of samples is chosen such that every shower would remain on average only once in the final sample after applying selection cuts. This balances effective use of the generated showers with the artificial fluctuations introduced by oversampling. Typically showers are re-sampled within radii ranging from 600 m to 2900 m for energies above 10^5 GeV (usually no re-sampling at lower energies).

9.4.3. 'Thinning'

Thinning of simulated showers [37] is a method to reduce computation time and storage volume by reducing the number of particles contributing to the shower development. This is achieved by keeping only one particle out of all secondary particles originating from a particle interaction with energies below a threshold energy $E_{thin} = \epsilon \times E_p$, where ϵ is the 'thinning level'. The kept particle gets a weight assigned which is the sum of energies of the particles below the threshold from which it was selected, divided by the kept particle's energy. This ensures the conservation of energy, although not of the particle number. The kept particles can interact again, leading to a multiplication of weights. Optionally, to reduce the fluctuations induced by thinning, no more thinning is applied to daughters of that particle if a weight exceeds a maximum weight w_{max} .

For IceTop/IceCube simulations a thinning level of $\epsilon = 10^{-6}$ is used for primary energies up to $E_p = 10^{8.4}$ GeV. Above this energy the thinning level is calculated as $\epsilon = 273 \text{ GeV}/E_p$ so that muons capable of penetrating deep into the ice (defined as E > 273 GeV for vertical muons) are not assigned thinning weights. The weights are limited to $w_{max} = \epsilon \times E_p$.

For the detector simulation 'un-thinning' has to be performed as follows: a weighted particle which hits a predefined sampling area around a tank generates a number of clones, each with the same energy as the parent, which are uniformly distributed over the sampling area. The number of particles is given by the weight such that the energy remains conserved. Technically only those clones are generated which have a chance to hit the tank. The radius of the sampling area is optimized such that on average only one particle hits a tank.

9.4.4. Computing resources

Cosmic ray simulation is performed on different, geographically dispersed computer clusters which are controlled by a GRID-based system. The produced data are transferred to the data center at the University of Wisconsin, Madison and become available for further processing.

At present, simulated CORSIKA showers for coincident events occupy 145 TB which is a large fraction of the available storage. In Table 10 typical CPU times and data sizes for single CORSIKA showers are reported. Simulations of the highest energy showers in the range between 10^7 and 10^9 GeV can take up to 2 days for a single



Table 10: Some typical CPU times for shower generation with CORSIKA.

Figure 31: The curves show the average charge in PE per VEM as a function of time between July 2009 and April 2012 for all DOMs which have been operational at the given time. The upper curve is for high-gain DOMs and the lower for low-gain DOMs.

shower with resulting file sizes of up to 10 GB. Simulations in the range from 10^4 to 10^7 GeV result in file sizes of the order of hundreds of MB with computing time of the order of hours.

The detector simulation needs about 1/3 of the corresponding CORSIKA computing time and adds only 3% to the disk usage.

10. Performance

10.1. Tank properties and detector stability

The commissioning history of IceTop tanks is reported in Table 2. About seven years after the first tanks were commissioned the long-term stability of the tank and ice properties as well as of the electronics can be evaluated. In November 2008, two test tanks that had been deployed in November 2003 were inspected to confirm that the quality of the ice and the bonding between the DOMs and the ice were still good. More quantitatively the stability is demonstrated with Fig. 31 showing the average signal charge per VEM as a function of time. The high-gain DOMs do not exhibit an obvious trend. For the low-gain DOMs a drop of about 10% over the displayed 30 months or 4% per year cannot be excluded. The changes per VEM calibration have typically a standard deviation of $\pm(4-6)$ PE. The PMT gains are usually adjusted by not more than 15% (the gain precision is estimated to be about 10%). Only one DOM out of the total of 324 IceTop DOMs shows permanent failure. The uptime of the detector is typically larger than 99%.

10.2. Air shower detection in IceTop

Effective area. Air showers are reconstructed if 3 or more stations have triggered as shown in Fig. 29. For the sub-sample where at least 5 stations have triggered, Fig. 32 shows the effective area as a function of energy from simulations of the IceTop array as commissioned in 2010 (IT73). The reconstructed shower cores were required to lie within a fiducial area of about $5.2 \times 10^5 \text{ m}^2$ with a polygonal shape excluding the outer stations. The effective exploitable energy range stretches from about 300 TeV to 1 EeV with nearly 3×10^7 events per year observed between 300 and 400 TeV, 200 events per year above 300 PeV and about 20 events per year above 1 EeV in the zenith range from $\cos \theta > 0.8$. There is no indication of array saturation up to 1 EeV [54]. Including the 3 and 4 station events the energy range can be extended to lowest energies of about 100 TeV due to the in-fill stations.



Figure 32: Effective area for proton showers which triggered 5 or more stations as a function of energy for the IceTop array as commissioned in 2010 (IT73).



Figure 33: Energy resolution for proton showers which triggered 5 or more stations as a function of energy for the IceTop array as commissioned in 2010 (IT73).

Energy, direction and position resolutions. The obtained energy, direction and core position resolutions for some selected energies are summarized in Table 11. In Fig. 33 the energy resolution as a function of energy is shown for proton showers in the IT73 configuration (same data set as for Fig. 32). The corresponding plots for the core and direction resolutions are displayed in Fig. 28.

These resolutions were determined in simulations by comparing the reconstructed with the generated variables. In IceTop, resolutions can also be studied experimentally since the same events can be reconstructed independently by the sub-array of A tanks and that of B tanks. From the comparison one can obtain an experimental measure of the accuracy of reconstruction. The sub-array resolution analysis is in good agreement with the simulation results. However, it should be noted that this method is not sensitive to systematic errors common to the whole detector.

Signal fluctuations. The sub-array method can also be used to determine the signal fluctuations which have to be included in the likelihood fit as described in section 8.3. Signal fluctuations are caused by detection fluctuations and fluctuations in the shower development. Both together are measured by comparing signals in the two tanks of

Table 11: Summary of air shower resolutions obtained with IceTop IT73 for protons in a zenith angle range $\cos \theta > 0.82$. A resolution is defined such that 68% of the events deviate less than the listed value.

energy	log-energy	zenith [°]	core [m]
400 TeV	0.1	0.8	9.0
1 PeV	0.1	0.6	7.5
10 PeV	0.06	0.3	5.5
100 PeV	0.05	0.4	5.5



Figure 34: Separation of elements in the primary cosmic ray composition as obtained with a neural net algorithm. The plot shows for proton, iron and a helium-oxygen mixture the separation in terms of a neural network output together with the rms width of the corresponding distributions. From [34].

a station (with proper corrections for different core distances). The detection fluctuations can be separated from the shower fluctuations by running both DOMs of a tank at the same gain and comparing their signals. In [1] it was shown that the detection fluctuations are much smaller than the shower fluctuations confirming that the tank design provides adequate (or better) resolution.

10.3. Cosmic ray anisotropy

Cosmic ray anisotropies have been studied by IceCube with high energy muons in the deep ice [55, 56]. An analysis using IceTop alone is currently in progress [57]. While the declination angular range is more restricted than for IceCube (to less than about 60°) the much higher energy resolution due to the shower reconstruction allows finer energy binning than for the muons detected with the in-ice detector. Anisotropy measurements can be done in the full IceTop energy range given above; sensitivities to intensity changes of less than 10^{-3} can be reached at least up to 10 PeV. The angular resolution of better than 3° is sufficient for studies of anisotropy patterns on a scale above about 10° .

10.4. Primary composition

10.4.1. Primary composition from coincident events

A strength of IceCube is the possibility to measure high energy muons in the deep ice in coincidence with the shower reconstructed in IceTop. The full IceCube detector can cover the energy range from below the knee to EeV. Showers generated by primary cosmic rays in this energy range produce multiple muons with energy sufficient to reach the depth of IceCube. For a primary energy of 10^{15} eV, for example, proton-induced showers near the vertical produce on average about 10 muons with $E_{\mu} > 500$ GeV and iron nuclei produce about 20 such muons. For higher primary energies, the number of muons increases, and the multiplicity in showers generated by nuclei approaches asymptotically a factor of $A^{0.24}$ times the muon multiplicity of a proton shower [58], yielding about 2.6 for A = 56. As a consequence of the high altitude of IceTop, showers are observed near maximum so the detector has good energy resolution, which is important when the goal is to measure changes in composition as a function of energy.

The first analysis of such coincident data is described in [34]. The data set is constrained to a subarray of the detector and a relatively short time period (about 1 month). An example of the separation of elements in this analysis is shown in Fig. 34. The separation increases with energy and is in the order of the width of the distributions. The measurement of the average logarithmic mass, $\langle \ln A \rangle$, from such distributions as presented in [34] is still dominated by systematic uncertainties, mainly from energy reconstruction in IceCube from changing environmental condition in IceTop and the hadronic interaction models for the shower development. In future analyses we expect to reduce the systematic uncertainties and to improve the separability by refining the algorithms and including additional mass sensitive variables. Mass sensitive observables are, for example, muon counts at the surface (see below), shower shape variables or in-ice observables like stochastic energy loss. A combined evaluation of such variables will allow constraining models.



Figure 35: Muon counting in IceTop: Distribution of tank signals for various cuts on the signal expectation S_{exp} in the energy range between 1 and 30 PeV. The average number of muons per event can be statistically extracted by fitting the muon signal together with the background (the example shows a fit for $S_{exp} < 0.125$ VEM).



Figure 36: Limits on the ratio of diffuse gamma ray and cosmic ray fluxes from within 10° of the Galactic Plane. The magenta line is the 90% confidence level limit obtained from one year of data taken with the IC40/IT40 configuration. The limits are compared to an analysis by CASA-MIA experiment at lower energies [61] and to the projected sensitivity of 5 years data taking with the full IceCube detector. The projected sensitivity is given for the full covered energy range by the dashed blue line and for smaller energy bins by the blue points.

10.4.2. Primary composition: Muon counting

Muonic and electromagnetic signals in IceTop can in general not be distinguished. However, muons show up as a relatively constant signal of about 1 VEM at larger distances from the shower axis where the expectation value of a tank signal, S_{exp} , becomes small compared to a muon signal (S_{exp} is obtained from the fit to the lateral shower distribution). As shown in Fig. 35, the muon signal becomes more and more prominent when requiring smaller S_{exp} [59]. For $S_{exp} < 0.125$ VEM the figure illustrates that the number of muons can be well fitted. Comparing these muon numbers as a function of energy to simulations of different primary masses, an independent information on the mass composition is obtained.

10.4.3. Primary photons

IceCube can efficiently distinguish PeV gamma rays from the background of cosmic rays by exploiting inice signals coincident with an IceTop event as veto against hadronic showers. Gamma-ray air showers have a much lower muon content than cosmic ray air showers of the same energy. Candidate events are selected from those showers that lack a signal from a muon bundle in the deep ice. Results of one year of data, taken in the 2008/2009 season when the detector consisted of 40 strings and 40 surface stations (IC40/IT40) are presented in [60]. Figure 36 shows the IC40/IT40 limits on diffuse gamma ray fluxes together with a projected gamma-ray sensitivity of the final detector (IC86/IT81). The limits are compared to a CASA-MIA analysis at lower energies [61]. Point source fluxes have also been searched for and excluded at a level of about $10^{-18} - 10^{-17} \text{ cm}^{-2} \text{s}^{-1} \text{TeV}^{-1}$ depending on the location in the sky. The complete IceCube detector will be sensitive to a flux of more than an order



Figure 37: IceTop veto against cosmic ray background in the search for extremely high-energetic neutrinos studied using data from one month running in the IC79/IT73 configuration. a) The plot shows the time distribution of any SLC hit in IceTop relative to the shower front expected if the event was a cosmic ray event (see details in text). The different distributions are for in-ice tracks which would pass the nearest IceTop tank at the distances given in the legend. The vertical solid black lines define the on-time window (cosmic ray signal), the dashed red lines the off-time window (background from random IceTop hits). b) The plot shows for the > 500 m curve in a) the distribution of the number of hits in the on-time window (black) and the off-time window (red).

of magnitude lower, allowing the search for PeV extensions of known TeV gamma-ray emitters. IceCube/IceTop has currently the highest gamma ray sensitivity in the energy range between about 1 and 10 PeV.

10.5. IceTop activity as veto for IceCube events

The production of extremely high-energetic (EHE) neutrinos is expected in collisions of cosmic rays with the cosmic microwave background. The detection of these cosmogenic neutrinos with IceCube [62] requires the ability to discriminate very rare and energetic signal events from an abundant background of cosmic ray induced muons. High energy cosmic ray air showers produce high numbers of muons densely packed around the shower core trajectory. These bundles of muons emit large amounts of Cherenkov light in the IceCube detection volume.

In [63] a study is presented which explores the possibilities of using the IceTop array to improve background rejection while keeping a large fraction of the neutrino signal. In a limited zenith range around the vertical this is obviously possible with high efficiency. However, by exploiting the IceTop SLC hits (single tank hits without additional conditions, see Section 4.2.2) air showers can also be efficiently rejected at larger zenith angles where the shower core does not hit IceTop.

The plots in Fig. 37 include data corresponding to 33.4 days from runs distributed over the whole year 2010 (IC79/IT73 configuration). Events were selected as neutrino candidates if they had more than 10^4 photons detected in IceCube, corresponding to a neutrino energy of more than 10^{17} eV. Under the assumption that the candidate is a cosmic ray event, a hypothetical shower front can be constructed from the in-ice track and its movement over the IceTop array can be followed. The plot in Fig. 37 a shows the time distribution of any SLC hit in IceTop relative to the expected shower front. The different distributions are for in-ice tracks which would pass the nearest IceTop tank at different distances of the shower core to the nearest IceTop tank. With the condition that the track passes through the in-ice detector the distances of the shower axes from the nearest IceTop tank correspond to zenith angle ranges: 250 m, 500 m and 750 m correspond to approximately 10-30°, 30-50° and 25-60°, respectively. The vertical solid black lines define the on-time window containing the cosmic ray signal, the dashed red lines the off-time window which yields the background of random hits in IceTop. The plot in Fig. 37 b shows for the > 500 m curve in a) the distribution of the number of hits in the on-time window (black) and the off-time window (red).

In the off-time window the hit multiplicity distribution drops very quickly, while the distribution of coincident hits is much flatter. The off-time hit distribution can be well described by a Poisson distribution with a mean of about $0.25/\mu$ s. That means requiring for a neutrino event to have less than 1 or 2 SLC hits yields neutrino signal reductions by 22% and 2.6%, respectively. The veto efficiency depends strongly on the required amount of light detected in IceCube and on the the distance of the extrapolated track to the nearest IceTop tank. From Fig. 37 b, where more than 10⁴ photons in IceCube and a closest approach of more than 500 m are required, one can read off a veto efficiency of about 75% for less than 2 SLC hits. Requiring 10⁵ photons in IceCube, the same criteria result



Figure 38: The difference between the zenith angle defined by the line connecting triggered IceTop station with the the center-of-gravity of triggered in-ice DOMs and the track reconstructed by the in-ice detector. In IceTop it was required that only one station has triggered and that the signal is made up by more than 400 photoelectrons.

in a veto efficiency well above 98%. Exploiting the IceTop veto the cuts on neutrino candidates can be widened gaining some 10% overall acceptance.

10.6. Calibration of IceCube with IceTop

Events reconstructed with IceTop that are also seen in the deep detector can be used to verify event reconstruction and the consistency of the relative alignment of both detector components. Examples of verification of timing and direction with IceTop are given in [1].

Showers that trigger IceTop produce usually bundles of several (at 1 PeV) or many muons in the deep detector. In contrast, much of the atmospheric muon background in deep IceCube consists of single muons. Most of these events are from cosmic-rays with primary energies below 10 TeV. By selecting a sample of coincident events in which both tanks at one and only one IceTop station are hit, it is possible to discriminate against high-energy events and find a sample enriched in single muons. Such a sample can be selected by the filter InIceSMT_IceTopCoin (Table 8 in Section 4.3.4). Coincidences involving only an interior IceTop station provide a sample in which about 75% are single muons in the deep detector [64].

An analysis of such coincident events using data taken in 2006 with the IC9/IT16 configuration is presented in [64]. Figure 38 shows the distribution of the differences between the zenith angle defined by the line from the hit station to the center-of-gravity of hits in the deep detector and the direction obtained from the muon reconstruction algorithm in the deep detector. The estimated angular accuracy of the connecting line is about 3° . The observed rms of the zenith difference distribution of 3.4° means that the in-ice reconstruction of muons has an accuracy of better than 2° . The mean of the distribution being close to 0 confirms the accuracy of positioning of both detectors as quoted in Section 2.6.

10.7. Heliospheric physics with IceTop

The IceTop tanks detect secondary particles from multi-GeV cosmic rays with a counting rate exceeding 1 kHz per detector. The detector response to secondary cosmic rays is discussed in [18].

A certain class of transient events, like solar flares or gamma ray bursts, can be detected with very good time resolution via a common rate increase in all IceTop tanks. The observation of the Dec 13, 2006 Sun flare event [65] demonstrates these abilities of IceTop. Following this observation the detector readout has been setup such that counting rates could be obtained at different thresholds (Section 3.3.3) allowing the unfolding of cosmic ray spectra during a flare (see Fig. 14).

11. Summary and Outlook

The IceTop air shower array, the surface component of the IceCube Neutrino Observatory, is operating in its final configuration since the beginning of 2011. In all aspects it has reached design performance. With an in-fill array the energy threshold is even lower than in the original design, leading to an energy coverage from about 100 TeV to 1 EeV. The detector is primarily designed to study the mass composition of primary cosmic rays by exploiting the correlation between the shower energy measured in IceTop and the energy deposited by muons in the deep ice.

First results on the energy spectrum and mass composition have been obtained with the still incomplete detector. We look forward to composition results from the complete detector. These results should lead to a substantial improvement of our understanding of cosmic rays in an energy range where the transition from galactic to extragalactic origin is expected.

In addition, the IceTop array has found various other applications. Counting rates of individual tanks as a function of different thresholds allow investigation of transient events from the sun and maybe other astrophysical objects. Using single tank hits IceTop can be very efficiently used for vetoing cosmic rays in the search for highenergy neutrinos. IceTop is also employed for direction and time calibration of the in-ice detector.

For the future an extension with radio antennas is being studied to enhance the air shower detection capabilities [66]. With the measurement of the shower development in air a radio detector would add complementary observables sensitive to the composition of cosmic rays.

Acknowledgments

We acknowledge the support from the following agencies: U.S. National Science Foundation-Office of Polar Programs, U.S. National Science Foundation-Physics Division, University of Wisconsin Alumni Research Foundation, the Grid Laboratory Of Wisconsin (GLOW) grid infrastructure at the University of Wisconsin - Madison, the Open Science Grid (OSG) grid infrastructure; U.S. Department of Energy, and National Energy Research Scientific Computing Center, the Louisiana Optical Network Initiative (LONI) grid computing resources; National Science and Engineering Research Council of Canada; Swedish Research Council, Swedish Polar Research Secretariat, Swedish National Infrastructure for Computing (SNIC), and Knut and Alice Wallenberg Foundation, Sweden; German Ministry for Education and Research (BMBF), Deutsche Forschungsgemeinschaft (DFG), Research Department of Plasmas with Complex Interactions (Bochum), Germany; Fund for Scientific Research (FNRS-FWO), FWO Odysseus programme, Flanders Institute to encourage scientific and technological research in industry (IWT), Belgian Federal Science Policy Office (Belspo); University of Oxford, United Kingdom; Marsden Fund, New Zealand; Australian Research Council; Japan Society for Promotion of Science (JSPS); the Swiss National Science Foundation (SNSF), Switzerland.

References

- [1] A. Achterberg *et al.*, IceCube Collaboration, "First year performance of the IceCube neutrino telescope", *Astropart. Phys.* **26** (2006) 155–173.
- [2] H. Kolanoski, for the IceCube Collaboration, "IceCube Astrophysics and Astroparticle Physics at the South Pole", in *Proceedings of the 32nd International Cosmic Ray Conference, Beijing*, Highlight talk given at 32nd International Cosmic Ray Conference, Beijing, China (ICRC 2011). 2011. arXiv:astro-ph.HE/1111.5188.
- [3] T. K. Gaisser, for the IceCube Collaboration, "IceCube: Status and Results", in Proceedings of the 14th International Workshop on Neutrino Telescopes, Venice, M. Mezzetto, ed., pp. 349-364. 2011. arXiv:astro-ph.HE/1108.1838.
- [4] M. A. Lawrence, R. J. O. Reid, and A. A. Watson, "The Cosmic ray energy spectrum above 4 × 10¹⁷ eV as measured by the Haverah Park array", J. Phys. G17 (1991) 733–757.
- [5] J. Abraham et al., Pierre Auger Collaboration, "Properties and performance of the prototype instrument for the Pierre Auger Observatory", Nucl. Instrum. Meth. A523 (2004) 50–95.
- [6] J. Blümer, R. Engel, and J. Hörandel, "Cosmic Rays from the Knee to the Highest Energies", Prog. Part. Nucl. Phys. 63 (2009) 293-338, arXiv:astro-ph.HE/0904.0725.
- [7] K. Nakamura et al., Particle Data Group, "Review of particle physics", J. Phys. G37 (2010) 075021.
- [8] K.-H. Kampert and M. Unger, "Measurements of the Cosmic Ray Composition with Air Shower Experiments", Astropart. Phys. 35 (2012) 660–678, arXiv:astro-ph.HE/1201.0018.
- [9] Silbrico Corporation, "RYOLEX Technical Data web page". http://www.silbrico.com/ryolex.html.
- [10] R. Abbasi et al., IceCube Collaboration, "The IceCube data acquisition system: Signal capture, digitization, and timestamping", Nucl. Instrum. Meth. A601 (2009) 294–316, arXiv:0810.4930.
- [11] R. Abbasi et al., IceCube Collaboration, "Calibration and Characterization of the IceCube Photomultiplier Tube", Nucl. Instrum. Meth. A618 (2010) 139–152, arXiv:astro-ph.IM/1002.2442.
- [12] Hamamatsu Corporation, "R7081-02 Photomultiplier Tube for IceCube Experiment". Data sheet, version 4, 2003.
- [13] S. Kleinfelder, "Gigahertz waveform sampling and digitization circuit design and implementation", *IEEE Trans. Nucl. Sci.* **50** no. 4, (2003) 955 962.
- [14] C. Roucelle, "DOMsimulator user's guide for IceSim release V02-00-00 and more...", Internal Report 200709002-v2, IceCube, 2007.
- [15] C. David, "Study of the IceCube DOM analog signals", Internal Report 200910002, IceCube, 2009.
- [16] T. Waldenmaier, "IceTop Configurations", Internal Report, IceCube, 2010.
- http://wiki.icecube.wisc.edu/index.php/IceTop_Configurations. IceCube Wiki, retrieved July 16, 2011.
- [17] T. Takao Kuwabara and P. Evenson, for the IceCube Collaboration, "Study of Forbush Decreases with IceTop", in *Proceedings of the* 32nd International Cosmic Ray Conference, Beijing, China (ICRC 2011). 2011. arXiv:astro-ph.HE/1111.2735.
- [18] J. M. Clem, P. Niessen, and S. Stoyanov, for the IceCube Collaboration, "Response of IceTop tanks to low-energy particles", in Proceedings of the 30th International Cosmic Ray Conference (ICRC 2007), Merida, Yucatan, Mexico. 2007. arXiv:astro-ph/0711.0353.

- [19] C. Roucelle, "Documentation for the DOMcalibrator module", Internal Report 200709001, IceCube, 2007.
- [20] T. K. Gaisser, "Semi-analytic approximations for production of atmospheric muons and neutrinos", *Astroparticle Physics* **16** no. 3, (2002) 285 294.
- [21] M. Beimforde, "Calibration of air shower signals in the IceTop detector using cosmic ray muons", diploma thesis, Humboldt-Universität zu Berlin, 2006.
- [22] ROOT Development Team, "ROOT web page". http://root.cern.ch/drupal/.
- [23] L. Demirörs et al., for the IceCube Collaboration, "IceTop tank response to muons", in Proceedings of the 30th International Cosmic Ray Conference (ICRC 2007), Merida, Yucatan, Mexico. 2007. arXiv:astro-ph/0711.0353.
- [24] X. Bai, T. Gaisser, A. Karle, K. Rawlins, G. Spiczak, et al., "Muon flux at the geographical south pole", Astropart. Phys. 25 (2006) 361–367, arXiv:astro-ph/0602381.
- [25] S. Tilav et al., for the IceCube Collaboration, "Atmospheric Variations as observed by IceCube", in Proceedings of the 31th International Cosmic Ray Conference (ICRC 2009), Lodz, Poland. 2009. arXiv:astro-ph.HE/1001.0776.
- [26] F. Kislat, Measurement of the Energy Spectrum of Cosmic Rays with the 26 Station Configuration of the IceTop Detector. PhD thesis, Humboldt-Universität zu Berlin, 2012. http://edoc.hu-berlin.de/docviews/abstract.php?lang=ger&id=39081.
- [27] R. Abbasi *et al.*, IceCube Collaboration, "All-particle cosmic ray energy spectrum measured with 26 IceTop stations", arXiv:astro-ph.HE/1202.3039. Submitted to ApP.
- [28] K. Kamata and J. Nishimura, "The Lateral and Angular Structure Functions of Electron Showers", *Progr. Theor. Phys. Suppl.* 6 (1958) 93–155.
- [29] K. Greisen, "Cosmic Ray Showers", Annu. Rev. Nucl. Sci. 10 (1960) 63-108.
- [30] S. Klepser, F. Kislat, H. Kolanoski, P. Niessen, and A. Van Overloop, for the IceCube Collaboration, "Lateral distribution of air shower signals and initial energy spectrum above 1-PeV from IceTop", in *Proceedings of the 30th International Cosmic Ray Conference (ICRC 2007), Merida, Yucatan, Mexico.* 2007.
- [31] S. Klepser, *Reconstruction of Extensive Air Showers and Measurement of the Cosmic Ray Energy Spectrum in the Range of 1-80PeV at the South Pole.* PhD thesis, Humboldt-Universität zu Berlin, 2008.
- http://edoc.hu-berlin.de/dissertationen/klepser-stefan-2008-06-24/.
- [32] F. Kislat, "Study of charge and time fluctuations of signals in the IceTop detector", diploma thesis, Humboldt-Universität zu Berlin, 2007. http://astro.desy.de/neutrino_astronomy/icecube/publications/diploma_theses/e597/ infoboxContent685/diplomarbeit.pdf.
- [33] K. Andeen, First Measurements of Cosmic Ray Composition from 1-50 PeV using New Techniques on Coincident Data from the IceCube Neutrino Observatory. PhD thesis, University of Wisconsin-Madison, 2011.
- [34] R. Abbasi *et al.*, IceCube Collaboration, "Cosmic Ray Composition from 1-30 PeV Using IceTop and IceCube", arXiv:astro-ph.HE/1207.3455. Submitted to ApP.
- [35] C. Xu, Chemical Composition of Primary Cosmic Rays with IceCube. PhD thesis, University of Delaware, Physics and Astronomy Department, 2012.
- [36] D. Heck et al., "CORSIKA: A Monte Carlo Code to Simulate Extensive Air Showers", Report FZKA, FZ Karlsruhe, 1998. http://www-ik.fzk.de/corsika/physics_description/corsika_phys.pdf.
- [37] D. Heck and T. Pierog, "Extensive Air Shower Simulation with CORSIKA: A Users Guide".
- http://www-ik.fzk.de/corsika/usersguide/corsika_tech.html.
- [38] S. Agostinelli et al., GEANT4 Collaboration, "GEANT4: A simulation toolkit", Nucl. Instrum. Meth. A506 (2003) 250-303.
- [39] J. Allison et al., "Geant4 developments and applications", IEEE Trans. Nucl. Sci. 53 (2006) 270.
- [40] R. S. Fletcher, T. K. Gaisser, P. Lipari, and T. Stanev, "SIBYLL: An event generator for simulation of high energy cosmic ray cascades", *Phys. Rev. D* 50 (1994) 5710–5731.
- [41] R. S. Fletcher, T. K. Gaisser, P. Lipari, and T. Stanev, "Air shower calculations with the new version of SIBYLL", in *Proceedings of the 26th International Cosmic Ray Conference*, vol. 1, p. 415. Salt Lake City, USA, 1999.
- [42] S. Ostapchenko, "QGSJET-II: towards reliable description of very high energy hadronic interactions", Nucl. Phys. B Proc. Suppl. 151 (2006) 143–146.
- [43] S. Ostapchenko, "QGSJET-II: results for extensive air showers", Nucl. Phys. B Proc. Suppl. 151 (2006) 147-150.
- [44] A. Fassò, A. Ferrari, J. Ranft, and P. Sala, "FLUKA: a multi-particle transport code", technical report, CERN, 2005. CERN-2005-10, INFN/TC_05/11, SLAC-R-773.
- [45] G. Battistoni et al., "The FLUKA code: Description and benchmarking", in Proceedings of the Hadronic Shower Simulation Workshop 2006, M. Albrow and R. Raja, eds., vol. 896 of AIP Conference Proceeding, pp. 31–49. 2007.
- [46] W. Nelson, H. Hirayama, and D. Rogers, "The EGS4 Code System", report slac-0265, SLAC, 1985. http://www.slac.stanford.edu/pubs/slacreports/slac-r-265.html.
- [47] A. M. Hillas, "Shower simulation: Lessons from MOCCA", Nucl. Phys. Proc. Suppl. 52B (1997) 29-42.
- [48] A. E. Hedin, "Extension of the MSIS thermosphere model into the middle and lower atmosphere", J. Geophys. Res. 96 (1991) 1159–1172.
- [49] T. Melzig, "Verbesserung der IceTop-Detektorsimulation unter Verwendung von Geant4", Master's thesis, in German, Humboldt-Universität zu Berlin, 2011. http://nuastro-zeuthen.desy.de/neutrino_experiments/icecube/ publications/diploma_theses/e597/infoboxContent110285/melzig_final_opt-1.pdf.
- [50] A. Van Overloop, for the IceCube Collaboration, "Simulation of IceTop VEM calibration and the dependency on the snow layer", in Proceedings of the 32nd International Cosmic Ray Conference, Beijing, China (ICRC 2011). 2011. arXiv:astro-ph.HE/1111.2735.
- [51] R. Glasstetter *et al.*, for the KASCADE Collaboration, "Analysis of electron and muon size spectra of EAS", in *Proc. 26th International Cosmic Ray Conference*. Salt Lake City, USA, 1999.
- [52] J. Hörandel, "Models of the knee in the energy spectrum of cosmic rays", Astropart. Phys. 21 (2004) 241–265.
- [53] T. K. Gaisser, "Spectrum of cosmic-ray nucleons, kaon production, and the atmospheric muon charge ratio", Astroparticle Physics 35 no. 12, (2012) 801 – 806, arXiv:astro-ph.HE/1111.6675.
- http://www.sciencedirect.com/science/article/pii/S0927650512000497.
- [54] T. Feusels and S. Tilav, for the IceCube Collaboration, "Extensive Air Showers Measured by the 79-string IceCube Observatory at South Pole", in *Proceedings of the 32nd International Cosmic Ray Conference, Beijing, China (ICRC 2011).* 2011. arXiv:astro-ph.HE/1111.2735.
- [55] R. Abbasi et al., IceCube Collaboration, "Measurement of the Anisotropy of Cosmic Ray Arrival Directions with IceCube",

Astrophys.J. 718 (2010) L194, arXiv:astro-ph.HE/1005.2960.

- [56] R. Abbasi et al., IceCube Collaboration, "Observation of an Anisotropy in the Galactic Cosmic Ray arrival direction at 400 TeV with IceCube", Astrophys.J. 746 (2012) 33, arXiv:hep-ex/1109.1017.
- [57] M. Santander, for the IceCube Collaboration, "Anisotropy of TeV and PeV cosmic rays with IceCube and IceTop", in Proceedings of the 2011 Very Large Volume Neutrino Telescopes Conference. 2012. arXiv:astro-ph.HE/1205.3969.
- [58] T. Gaisser, *Cosmic Rays and Particle Physics*. Cambridge University Press, 1990.
 [59] H. Kolanoski, for the IceCube Collaboration, "The IceTop Air Shower Array: detector overview, physics goals and first results", in Proceedings of the 32nd International Cosmic Ray Conference, Beijing, China (ICRC 2011). 2011. arXiv:astro-ph.HE/1111.2735. [60] S. Buitink, for the IceCube Collaboration, "Searching for PeV gamma rays with IceCube", in proceedings of the 32nd International
- Cosmic Ray Conference, Beijing, China (ICRC 2011). 2011. arXiv:astro-ph.HE/1111.2735. paper 939. [61] M. Chantell et al., CASA-MIA Collaboration, "Limits on the isotropic diffuse flux of ultrahigh-energy gamma radiation",
- Phys.Rev.Lett. 79 (1997) 1805-1808, arXiv:astro-ph/9705246.
- [62] R. Abbasi et al., IceCube Collaboration, "The first search for extremely-high energy cosmogenic neutrinos with the IceCube Neutrino Observatory", Phys. Rev. D82 (2010) 072003, arXiv:astro-ph.CO/1009.1442.
- [63] J. Auffenberg et al., for the IceCube Collaboration, "New background rejection methods for the GZK neutrino search with IceCube", in Proceedings of the 32nd International Cosmic Ray Conference, Beijing. 2011. arXiv:astro-ph.HE/1111.2736.
- [64] X. Bai, T. Gaisser, T. Stanev, and T. Waldenmaier, for the IceCube Collaboration, "IceTop / IceCube coincidences", in Proceedings of the 30th International Cosmic Ray Conference (ICRC 2007), Merida, Yucatan, Mexico. 2007. arXiv:astro-ph/0711.0353.
- [65] R. Abbasi et al., IceCube Collaboration, "Solar Energetic Particle Spectrum on 13 December 2006 Determined by IceTop", Astrophysical Journal Letters 689 (2008) 65-68.
- [66] M. DuVernois, for the IceCube Collaboration and ARA Collaboration, "The Radio Air Shower Test Array (RASTA) enhancing the IceCube observatory", in Proceedings of the 32nd International Cosmic Ray Conference, Beijing. 2011. arXiv:astro-ph.IM/1111.2742.